THE JOURNAL OF FINANCE • VOL. LXXIV, NO. 4 • AUGUST 2019

Income Hedging, Dynamic Style Preferences, and Return Predictability

JAWAD M. ADDOUM, STEFANOS DELIKOURAS, GEORGE M. KORNIOTIS, and ALOK KUMAR*

ABSTRACT

We propose a theoretical measure of income hedging demand and show that it affects asset prices. We focus on the value factor and first demonstrate that our demand estimates are correlated with the actual demands of retail and mutual fund investors. We then show that the aggregate high-minus-low (HML) demand predicts HML returns. Exploiting the state-level variation in income risk, we demonstrate that state-level hedging demands predict state-level HML returns. A long-short portfolio that exploits this hedging-induced predictability earns an annualized risk-adjusted return of 6%.

INCOME RISK IS A FUNDAMENTAL source of uncertainty that households face, and hence it should affect their financial decisions. Several recent studies demonstrate that income risk influences the portfolio decisions of U.S. and European households. For example, Angerer and Lam (2009) demonstrate that U.S. households with higher permanent income risk invest less in risky assets. Betermier et al. (2012) show that the financial decisions of Swedish households are sensitive to wage volatility.

A related literature provides household-level evidence that portfolio decisions are affected by income hedging considerations. Bonaparte, Korniotis, and Kumar (2014) find evidence of income hedging in the decisions of Dutch and U.S. households. They show that households whose income growth is highly correlated with market returns participate less in the market and allocate less of their wealth to risky assets. Using Swedish data, Betermier, Calvet, and Sodini (2017) argue that the tilt of some investors toward value versus growth stocks is motivated by income hedging concerns. They show that income

*Jawad M. Addoum is with S.C. Johnson College of Business, Cornell University. Stefanos Delikouras, George M. Korniotis, and Alok Kumar are with University of Miami Business School. We thank three anonymous referees; an anonymous Associate Editor; Sandro Andrade; Tim Burch; Sean Campbell; Vidhi Chhaochharia; Jason Delaney; Andrew Karolyi; David Ng; Tarun Ramadorai; Kenneth Singleton (Editor); Luis Viceira; Deniz Yavuz; and seminar participants at Cornell University, the Federal Reserve Board in Washington DC, Florida International University, Goethe University Frankfurt, University of Mannheim, University of Miami, University of Warwick, the 2016 EFA Annual Meeting, NTA 106th Annual Conference on Taxation, and 2016 SFS Finance Cavalcade for helpful comments and valuable suggestions. We are responsible for all remaining errors and omissions. The authors have no conflicts of interest as identified by the *Journal of Finance* Disclosure Policy.

DOI: 10.1111/jofi.12775

hedging considerations are strongest for those households with high human capital and high exposure to aggregate risk.¹ Korniotis and Kumar (2011) further demonstrate that certain types of financial assets facilitate income smoothing and improve risk sharing across U.S. states.

Motivated by the household-level evidence on income hedging, in this paper, we examine its potential asset pricing implications. In particular, we investigate the potential link between the aggregate income hedging behavior of U.S. households and the predictability in stock returns. Our primary conjecture is that the demand for financial assets that facilitate income hedging varies as their hedging potential changes over time. If such hedging-induced demand shifts are systematic, they could affect prices. This conjecture is partially motivated by dynamic portfolio choice models, which demonstrate that when investment opportunities are time-varying, investors should rebalance their portfolios as they learn about future investment opportunities (e.g., Campbell and Viceira (1999), Campbell and Vuolteenaho (2004), and Jurek and Viceira (2011)).

We connect this basic prediction of dynamic portfolio choice models with the literature on style investing, which shows that investors systematically move in and out of certain investment styles (e.g., Barberis and Shleifer (2003), Kumar (2009), and Wahal and Yavuz (2013)). We argue that style shifts between value and growth stocks might be motivated in part by the time-variation in income hedging opportunities. Such systematic demand shifts could generate predictable variation in the returns of value and growth portfolios.

We formalize this economic insight in a model that combines the noise trader specification of De Long et al. (1990), the Bayesian framework of Barberis (2000), and the income hedging model of Viceira (2001). In the model, there are two groups of investors. The first group comprises workers, who receive stochastic income and are concerned about income risk. The second group is the sophisticated investors, who are risk neutral and face transaction costs. We solve the model analytically, and obtain explicit solutions for the optimal portfolios of workers and sophisticated investors. Similar to Viceira (2001), asset demands by workers include an income hedging term, which depends on the covariances between asset returns and income, scaled by the covariance matrix of asset returns. This covariance-based term is our proposed measure of income hedging demand (IHD).

When the conditional covariance between income and an asset decreases, the income hedging potential of this asset increases and the IHD measure increases. Workers then demand more of the asset, creating price pressure, which in our model cannot be fully absorbed by the sophisticated investors due to transaction costs. This price pressure today generates lower returns in the future. Overall, the key theoretical prediction is that there is a *negative* relation between our IHD measure and future asset returns.

¹ Other papers related to income hedging include Heaton and Lucas (2000), Viceira (2001), and Vissing-Jorgensen (2002).

To test our predictability hypothesis, we focus on the value factor or the high-minus-low (HML) style portfolio. We focus on the HML factor because existing literature suggests that income risk affects the choice between value and growth stocks. For example, studies using aggregate data find that the value premium is related to income hedging. In particular, Koijen, Lustig, and Nieuwerburgh (2017) show that value stocks are especially risky because their cash flows are low during deep recessions, that is, during periods when aggregate income growth is very negative. Thus, investors could hedge against this risk by investing in growth stocks.

Betermier, Calvet, and Sodini (2017) provide direct evidence of income hedging using household-level data. They find that households with high income risk or high human capital and households employed in highly cyclical sectors avoid value stocks and prefer growth stocks. Cronqvist, Siegel, and Yu (2015) also find that value versus growth investing is one of the most predominant investment styles. Further, they show that households with higher human capital (i.e., high labor income and high education) tend to prefer growth stocks. Moreover, investors with more procyclical income (i.e., high correlation between labor income growth and GDP growth) also tend to prefer growth stocks.

We also focus on the HML portfolio because it satisfies the necessary conditions required to generate predictability in returns. First, value and growth investment styles have been popular as far back as the 1930s (e.g., Graham and Dodd (1934)). Further, the average investor can easily gain exposure to value and growth portfolios (Wahal and Yavuz (2013)). Jurek and Viceira (2011) report that in the universe of mutual funds in the Center for Research in Security Prices (CRSP) database, about 78% of funds can be categorized as either value or growth funds. Second, the income hedging potential of the HML portfolio varies significantly over time, which generates time-varying demand shifts that can potentially influence stock prices.

Since hedging-induced investments cannot be directly observed, we estimate the IHD using the relation predicted by our model. Because we want to examine the aggregate asset pricing implications of income hedging, we construct the IHD using aggregate U.S. and state-level income data. Specifically, we adopt a conditional estimation approach where we compute variances and covariances using a 10-year rolling window. Using these conditional moment estimates we compute IHD at both the U.S. state level and the national level. As shown in Figure 1, the IHD estimates exhibit substantial variation in the cross section and over time. For instance, in every period some U.S. states have negative while others have positive IHD estimates.

Before using our IHD measure to predict asset returns, we investigate whether these theoretical demand estimates are related to actual investor demands for value and growth stocks. We perform two validation tests. In the first test, we compute the relative portfolio weights in value and growth stocks based on the actual stock holdings of a sample of retail investors. We find that the correlation between the actual relative weight and our IHD measure is positive and statistically significant.



Figure 1. Time-variation in income hedging demand (IHD). This figure displays the time series of IHD. Panel A displays the time series of the average state-level IHD (dark line) over the sample period. We also show the range spikes extending from the 10th to the 90th percentile of the cross-sectional distribution of the state-level IHD each year (light vertical lines). The state-level IHD is based on scaled conditional covariances between state-level income and HML, and then averaged each quarter across states, weighting each state by the fraction of national financial wealth held by state residents (calculated using IRS data on capital gains and dividend income). Conditional covariances are calculated as the covariance between changes in quarterly state-level real income per-capita and HML returns over the past 40 quarters for each quarter during the sample period with 40 trailing observations. In Panel B, IHD is based on conditional covariances between changes in quarterly U.S. real income per-capita and HML returns over the past 40 quarters for each quarter during the sample period with 40 trailing observations. Also shown in Panel B are the two-standard-error bands around the sample mean of the IHD (dashed lines). The sample period is from Q1 1970 to Q4 2011. (Color figure can be viewed at wileyonlinelibrary.com)

Income Hedging, Dynamic Style Preferences, and Return Predictability 2059

In the second validation test, we use mutual fund data from CRSP. Consistent with our model, we find that the average net flows into value funds are positively correlated with our IHD measure. In contrast, the average net flows into growth funds are negatively correlated with our IHD measure. These findings suggest that the theoretical covariance-based demand measures capture the portfolio decisions of actual market participants reasonably well.

In our asset pricing tests, we first focus on aggregate U.S. income risk and show that our U.S.-level IHD estimates can predict HML portfolio returns. We find that high IHD is associated with low HML returns in the next period. In economic terms, a one-standard-deviation increase in the IHD is associated with 0.59% lower HML return next month and 0.96% lower HML return next quarter. The evidence of predictability remains significant for about a year, suggesting that shifts in hedging-induced demand have a medium-term impact on asset prices. These findings are robust to a number of control variables, and they are not driven by macroeconomic shocks or estimation biases.

Next, we examine the implications of income hedging for asset prices across U.S. states. Consistent with our previous evidence, we find that state-level IHD estimates can predict state-level HML returns, especially among stocks with high local ownership. In contrast, among stocks with low local ownership, evidence of hedging-induced return predictability is weak. A long-short portfolio that exploits hedging-induced predictability in the cross section of U.S. states earns an annualized risk-adjusted return of over 6%. This evidence suggests that at least part of the HML predictability at the U.S. state level is induced by geographical heterogeneity in income risk.

We complement our main results with a set of tests designed to show that our main findings are related to income hedging. First, we follow Davis and Willen (2000), who show that across 10 occupation categories, electrical engineers face the highest income risk and auto mechanics face the lowest. Given their findings, we expect the IHD of electrical engineers, who have the most to gain from income hedging, to more closely replicate our main result than that of auto mechanics, who are unlikely to exhibit income hedging behavior. Consistent with this conjecture, we find that the electrical engineers' IHD measure is a statistically significant predictor of HML returns. In contrast, auto mechanics' IHD is an insignificant predictor of HML returns.

Second, Guvenen et al. (2017) suggest that income hedging motives vary across households due to cross-sectional differences in the exposure to aggregate risk. They show that those at the very bottom and very top of the income distribution are the ones who bear income risk both with respect to aggregate GDP and their employer's earnings. Since most market participants are at the top of the income distribution (Campbell et al. (2016)), the findings in Guvenen et al. (2017) suggest that the IHD of the top earners should have the most predictive power for returns. Consistent with this view, we find that the most important IHD predictors are those calculated using income growth for the 96% to 99% and top 1% of households.

Collectively, our findings contribute to the literature on household portfolio choice and income hedging (e.g., Samuelson (1969), Merton (1969, 1971, 1973),

Viceira (2001)). Specifically, we propose a measure of aggregate IHD, which we validate using data on retail investors and mutual fund flows. Our measure is theoretically motivated by a structural model with two types of investors (workers and sophisticated investors), which we solve analytically.

Further, our paper contributes to the literature on style investing. Barberis and Shleifer (2003) posit that investors' grouping of assets into broad styles can help explain excess comovement within style categories (Lee, Shleifer, and Thaler (1991), Barberis, Shleifer, and Wurgler (2005)). Teo and Woo (2004) demonstrate that style investing is related to momentum and reversal patterns. Kumar (2009) and Campbell, Ramadorai, and Ranish (2014) document styleswitching behavior among retail investors. Kumar (2009) shows that styleswitching behavior has an effect on stock returns.

More recently, Wahal and Yavuz (2013) show that past style returns are significant predictors of future returns for individual stocks categorized into styles. Specifically, they test whether style investing leads to return predictability by examining whether past style returns have any predictive power in the cross section. Similar to our empirical framework, they identify styles using size and book-to-market portfolios. However, they directly test their conjecture using Fama and MacBeth (1973) regressions of future stock returns on size, book-to-market, past stock returns, and past style returns. Compared to Wahal and Yavuz (2013), our starting point is an economic model in which IHD is a predictor of returns.

Overall, we connect the literatures on style investing and income hedging, and show that a driver of the style-switching activities of investors is likely to be their income hedging motives. Specifically, we show that income hedging generates systematic shifts in demand across style portfolios and can predict future style returns. This income-hedging-induced predictability mechanism is novel and has not been studied by existing asset pricing literature on human capital (e.g., Fama and Schwert (1977b), Campbell (1996), Jagannathan and Wang (1996), Campbell et al. (2016)).

Our work also complements recent literature on how firms insure workers through their labor contracts, which can affect asset prices (Berk and Walden (2013), Marfè (2017)). We complement this work and argue that both hedging via labor contracts and financial markets can coexist. Guvenen et al. (2017) show that those at the very bottom and very top of the income distribution are the ones who bear income risk. In contrast, workers in the middle of the income distribution are more likely to be insured by their employers, and hence do not bear much income risk. Consistent with this view, we show that our results are largely driven by the income risk of those at the top, and not the middle, of the income distribution.

Further, we present two tests suggesting that hedging via financial markets and labor contracts are likely to be complementary channels. First, we split U.S. states into those with high unionization and low unionization rates. Since unions typically advocate for rigid wage contracts (e.g., Shister (1943)), workers in areas with higher unionization rates should have a weaker motivation to selfinsure in financial markets. We find that our results hold in both subsamples. In our second test, we compute the labor share measure of Marfè (2017) and include it as an additional control variable in our national-level HML predictive regressions. We find that IHD remains an economically and statistically significant HML predictor in the presence of the labor share measure.

The rest of the paper is organized as follows. Section I discusses a model with time-varying IHDs. Section II describes our method for estimating hedging demand and provides estimates of IHDs. Section III presents validation tests comparing IHD estimates with actual time-variation in demand for value and growth stocks. Section V provides evidence of return predictability at the aggregate level and Section V extends the predictability analysis to the state level. Section VI includes results from additional analyses as well as robustness tests. Finally, Section VII concludes with a brief discussion.

I. A Two-Investor Model with Income Hedging

Our empirical analysis is based on a model that describes how time-variation in the income hedging potential of financial assets can generate predictability in asset returns. The model generates a theoretically motivated measure of income hedging, and highlights the conditions under which this measure should be related to future asset returns.

Our model extends the noise trader specification of De Long et al. (1990) to account for Bayesian learning as in Barberis (2000) and income hedging as in Viceira (2001). We follow this approach because most of the literature that seeks to explain return predictability uses the two-investor setup of De Long et al. with Bayesian learning. See, for example, Timmermann (1993, 1996), Lewellen and Shanken (2002), and Pastor and Veronesi (2009). The main purpose of our model is to provide a formal way for studying the relation between return predictability and income hedging.²

For our theoretical framework, we consider a closed endowment economy as in De Long et al. (1990). In this economy, there is a risk-free asset. Its supply is perfectly elastic and its price is fixed at one. The riskless asset pays interest $r_f, t + 1$, which is constant and positive $(r_{f,t+1} = r_f > 0)$. The *n* risky assets of the economy are in limited fixed supply. The fixed supply of risky assets is denoted by the $n \times 1$ vector \bar{N} . The vector p_t is the vector of prices for the risky assets, and d_t is the vector of dividends paid by the risky assets. For simplicity, we assume that dividends are i.i.d. with mean \bar{d} and covariance matrix Σ_d . The vector r_{t+1}^x denotes cum-dividend returns in excess of the risk-free rate and is equal to $p_{t+1} + d_{t+1} - (1 + r_f)p_t$. Finally, $\mu_{r,t} = \mathbb{E}_t[r_{t+1}^x]$ is the vector of risk premia, and the matrix $\Sigma_{r,t} = Var_t(r_{t+1}^x)$ is the conditional covariance matrix of cum-dividend returns.

The economy is populated by overlapping generations of investors who live for two periods. There are two types of investors: workers and sophisticated

² Our model is related to existing work on the relation between human capital and asset returns. For example, see Campbell (1996), Jagannathan and Wang (1996), and Campbell et al. (2016). None of these studies examine return predictability, which is our focus.

investors, with population masses ν and $1 - \nu$, respectively ($\nu \in (0, 1)$). In the first period, both types of investors receive exogenous endowments that can be invested in the risky and risk-free assets. There is no consumption in the first period. In the second period, investors receive dividends, sell their investments, consume all of their wealth, and die.

Sophisticated investors are risk-neutral. They have complete information and are characterized by rational expectations. In the first period, sophisticated investors pay quadratic transaction costs for investing in risky assets as in Gârleanu and Pedersen (2013). These costs are a dead-weight loss to the economy and proxy for limited external capital. The transaction costs are summarized by a diagonal matrix \boldsymbol{Q} . For simplicity, we assume that \boldsymbol{Q} is equal to $(q/2) \times \boldsymbol{I}$, where q is a positive scalar and \boldsymbol{I} is the identity matrix.³

Workers are characterized by quadratic utility with identical risk aversion parameters γ ($\gamma > 0$). In addition to their first-period endowment (y_t), during the second period workers receive another stochastic endowment (income), which is denoted by y_{t+1} . The dynamics of income are the same for all workers and across generations. Income dynamics are described by the following process with time-varying volatility:

$$y_{t+1} = y_t + \mu_y + \sigma_{y,t} e_{y,t+1}.$$
 (1)

The constant μ_y denotes expected income changes, and $e_{y,t+1}$ are i.i.d. normal shocks with zero mean and unit variance. The parameter $\sigma_{y,t}$ captures the time-variation in income volatility. This parameter is orthogonal to all other shocks in the economy. To ensure that the time-varying volatility is always positive, we assume that $\sigma_{y,t}$ follows an i.i.d. inverse-gamma distribution $IG(\xi, \mu_{\sigma_y}(\xi - 1))$ with $\xi > 2$ and $\mu_{\sigma_y} > 0$. The income process in equation (1) is quite standard in the consumption and income hedging literatures (e.g., Mankiw and Shapiro (1985)). The only difference relative to existing models is that we assume income heteroscedasticity with time-varying volatility. Moreover, following Viceira (2001), we assume that markets are incomplete. Therefore, the income risk of workers described in equation (1) cannot be perfectly hedged by the traded assets. However, asset returns and workers' income are conditionally correlated with $\Sigma_{y,r,t}$, the $n \times 1$ vector of conditional covariances between asset returns and income in first-differences.⁴

Moreover, we assume that workers are less informed than the sophisticated investors. Specifically, as in Barberis (2000), workers infer return and income

 3 Technically, the presence of transaction costs guarantees that the demand of the risk-neutral sophisticated investors is finite.

⁴ We remain agnostic as to why labor income and asset returns are correlated. Campbell et al. (2016) discuss this issue in relation to HML factor returns. They show that income growth of the top 1% of the income distribution is a priced factor that is related to the HML factor. They conjecture that high-income investors may be disproportionately employed by value rather than by growth firms. Using data from Compustat, they find that their high-income factor covaries more with the per-employee wage growth of value firms than with the wage growth of growth firms. This intuition can apply in part to our setting as well since we find that the IHD of the top 1% of the income distribution can predict the HML return. See Section VI.B.

moments using Bayes's rule with uninformative priors. Although workers live only for two periods, they observe a history of T observations for income and asset prices. To derive closed-form solutions for asset demands, we also assume that workers believe that asset returns *and* income changes are jointly normally distributed i.i.d. variables with unknown mean and volatility.

Finally, workers do not observe the demand of sophisticated investors, whereas sophisticated investors are fully aware of workers' decisions. This assumption is reasonable because sophisticated investors represent market makers and brokers, who in practice observe the limit order flow and can infer investors' decisions. Sophisticated investors can also be institutional investors and hedge fund managers who have more resources and better information. In contrast, it is reasonable to assume that workers, who represent the average retail investor, have access only to public information (like the history of asset prices) and generally have less resources to infer the decisions of institutional investors and form expectations.⁵

A. Market-Clearing: Hedging and Asset Prices

Based on the above assumptions, we can derive the market-clearing condition.

PROPOSITION 1: The market-clearing condition for our model economy is given by

$$\bar{\boldsymbol{N}} = \frac{1-\nu}{q(1+r_f)} \mathbb{E}_t^{so}[\boldsymbol{r}_{t+1}^x] + \frac{\nu(T-n-3)}{\gamma(T+1)} \widehat{\boldsymbol{\Sigma}}_{r,t}^{-1} \widehat{\boldsymbol{\mu}}_{r,t} + \nu IHD_t.$$
(2)

PROOF: See the Appendix.

The market-clearing condition (2) depends on three terms. The first is the vector of expected excess returns based on sophisticated investors' rational beliefs (\mathbb{E}_{t}^{so} []), which reflect complete information. When expected excess returns are high, sophisticated investors demand more of the risky assets. The second term captures the traditional risk-return trade-off based on workers' estimates of risk premia ($\hat{\mu}_{r,t}$) and asset variances ($\hat{\Sigma}_{r,t}$). When workers estimate high risk-return ratios, they demand more of the risky assets. The term (T - n - 3)/(T + 1) is an adjustment to the risk-return ratio because workers are Bayesian optimizers with uninformative priors.⁶ This term accounts for parameter uncertainty.

The third term in equation (2) is workers' estimated IHD. The IHD term is the *negative* value of the estimated covariances of asset returns and income

 $^{^5}$ There can be multiple groups of sophisticated investors. For tractability, we avoid introducing more than two types of investors, which is consistent with many existing models.

 $^{^{6}}$ T is the number of time-series observations that workers use to estimate return and income moments and n is the number of assets.

changes scaled by the estimated asset variances:

$$IHD_t = -\widehat{\Sigma}_{r\,t}^{-1}\widehat{\Sigma}_{y,r,t}.\tag{3}$$

When the covariances between asset returns and income in first-differences decrease, the income hedging potential of financial assets increases. Consequently, workers demand more of the risky assets. In contrast, when the estimated covariances between asset returns and income in first-differences increase, the income hedging potential of financial assets decreases. In this case, workers demand less of the risky assets.⁷

B. Empirical Predictions: Hedging and Return Predictability

We rearrange the market-clearing condition (2) in terms of sophisticated investors' expected returns and obtain the following expression:

$$\mathbb{E}_{t}^{so}[\boldsymbol{r}_{t+1}^{x}] = \frac{q(1+r_{f})}{1-\nu} \bar{\boldsymbol{N}} - \frac{\nu q(1+r_{f})}{1-\nu} \frac{T-n-3}{\gamma(T+1)} \widehat{\boldsymbol{\Sigma}}_{r,t}^{-1} \widehat{\boldsymbol{\mu}}_{r,t} - \frac{\nu q(1+r_{f})}{1-\nu} IHD_{t}.$$
 (4)

Since sophisticated investors have rational expectations, we can replace the expectation operator in equation (4) with an error term ϵ_{t+1} , to conclude that

$$\boldsymbol{r}_{t+1}^{x} = \frac{q(1+r_{f})}{1-\nu} \bar{\boldsymbol{N}} - \frac{\nu q(1+r_{f})}{1-\nu} \frac{T-n-3}{\gamma(T+1)} \widehat{\boldsymbol{\Sigma}}_{r,t}^{-1} \widehat{\boldsymbol{\mu}}_{r,t} - \frac{\nu q(1+r_{f})}{1-\nu} IHD_{t} + \boldsymbol{\epsilon}_{t+1}.$$
 (5)

Equation (5) makes a clear empirical prediction. It implies that changes in hedging demand lead to predictable patterns in returns. Specifically, when the income hedging potential of financial assets increases, IHD will also increase, and equation (5) suggests that future returns will be lower. Equation (5) also predicts a negative relationship between the term $\widehat{\Sigma}_{r,t}^{-1}\widehat{\mu}_{r,t}$ and future returns. We call this term the RISK-RETURN term and include it in our estimation for completeness.

Equation (5) clarifies the conditions under which income hedging can lead to return predictability. First, the effect of income hedging on returns depends on the magnitude of the transaction costs q, which proxies for costly capital. A large q implies that the effect of income hedging on returns is strong because

⁷ In the model, time-variation in IHD, and thus predictability in returns, arises because workers are Bayesian updaters who use past observations to estimate asset pricing moments. Hence, the market-clearing condition in equation (2) implies a high-order, nonlinear recursion for the pricing function, which does not have a closed-form solution. The nontractability of the pricing function is due to Bayesian updating by workers. We can obtain an explicit solution for the pricing function for a limiting case. Specifically, in Section I of the Internet Appendix, we assume that workers have complete information. In this case, time-variation in IHD is due to time-variation in income volatility. The complete information model of the Internet Appendix is very similar to that of De Long et al. (1990), with the exception that, instead of noise traders, we consider income earners who optimally hedge income risk. The Internet Appendix may be found in the online version of this article.

2064

sophisticated investors cannot trade easily due to high transaction costs. Second, the effect of income hedging on returns depends on the relative population masses of workers and sophisticated investors, that is, ν and $1 - \nu$, respectively. In particular, the effect of income hedging strengthens when there is a large number of workers and a small number of sophisticated investors in the economy. In sum, as long as investing in risky assets is costly and there is a positive mass of workers, shifts in workers' asset demands can generate predictability in asset returns.

C. Simulation Evidence and the Income Hedging Mechanism

Before estimating equation (5), we provide simulation evidence showing that for reasonable model parameters, the predictability generated by the model matches the predictability we document in our empirical tests. This is an important exercise that lends credence to our conjecture that the observed predictability is related to income hedging.

In particular, in Section II of the Internet Appendix, we simulate expected HML returns $(\mathbb{E}_t[\mathbf{r}_{t+1}^{HML}])$ for our two-investor model using the observed IHD measure and the market-clearing condition from equation (2). The simulation exercise shows that when we calibrate the deep economic parameters in equation (2) to match the findings in previous literature ($r_f = 1.26\%$, $\nu = 0.4$, q = 0.4%, $\gamma = 13$) and the structure of our empirical design (T = 40, n = 4), the observed IHD is able to generate the HML return predictability we see in the data. These results are reported in Table IA.I of the Internet Appendix.

II. IHD of Financial Assets

Our empirical analysis is based on the following key implication of our model: If the hedging potential of financial assets is time-varying, then IHD should also be time-varying, which can generate predictable patterns in asset returns. In this section, we examine whether there is time-variation in the income hedging potential of the value factor. We focus on the value factor for a variety of reasons, as explained in Section II.A. We also use aggregate income data because we want to measure the systematic effects of income hedging that arise at the aggregate market level.

A. Choice of HML as an Income Hedging Instrument

To operationalize our conjecture that changes in hedging demand can generate predictable patterns in stock returns, we need to identify portfolios that have substantial income hedging potential. In addition, to identify predictability in realized portfolio returns, it is necessary to focus on investment strategies that are well known and accessible to investors (e.g., Wahal and Yavuz (2013)). We focus on the value factor, or the HML portfolio, because it satisfies both of these requirements. To begin, the value effect has been documented as far back as the 1930s (e.g., Graham and Dodd (1934)). Also, as we see in Figure 1, the IHD of HML varies considerably over time. Further, the pervasive nature of value and growth suggests that investors can adopt these investment styles easily. For example, Jurek and Viceira (2011) report that in the universe of mutual funds in the CRSP database, about 78% of funds can be categorized as either value or growth funds.

Existing literature has also documented that income risk affects the choice between value and growth stocks. In particular, there is ample evidence that the HML premium is compensation for risk related to aggregate income.⁸ More recently, Koijen, Lustig, and Nieuwerburgh (2017) provide explicit evidence that value stocks are especially risky because their cash flows are low during deep recessions, that is, during periods when aggregate income growth is very negative. Thus, investors could hedge against this risk by investing in growth stocks. Overall, studies that use aggregate data find evidence consistent with the value premium being related to income hedging.

Betermier, Calvet, and Sodini (2017) use household-level data and provide direct evidence of income hedging. They find that households with high income risk or high human capital and households that are employed in highly cyclical sectors avoid value stocks and prefer growth stocks. They offer various explanations as to why the level of human capital in particular should affect the preference of value versus growth stocks. They argue that in production of asset pricing models (e.g., Berk, Green, and Naiko (1999), Zhang (2004)), human and physical capital are complementary factors of production. Thus, owners of high human capital should hedge income risk by not investing in firms with high physical capital, which tend to be value firms.

Similar to Betermier, Calvet, and Sodini (2017), Cronqvist, Siegel, and Yu (2015) show that value versus growth investing is a predominant investment style. These authors provide evidence of a genetic component to the preference of value and growth stocks. They also show that households with higher human capital (i.e., high labor income and high education) tend to prefer growth stocks. Moreover, investors with more procyclical income (i.e., a high correlation between labor income growth and GDP growth) also tend to prefer growth stocks.

A related literature offers another explanation as to why there might be a relation between income risk and the HML factor. Specifically, Kogan et al. (2017) build a production model based on the evidence that innovation at the firm level is an important determinant of future growth (Kogan, Papanikolaou, and Stoffman (2017)). In the model, firms that develop new technologies and innovate are growth firms, and the remaining firms are value firms. More importantly, the rents from innovation accrue to very few agents, while the rest of the population is exposed to creative destruction or displacement risk

⁸ See Jagannathan and Wang (1996), Campbell (1996), Jagannathan, Kubota, and Takehara (1998), Lettau and Ludvigson (2001b), Lustig and Nieuwerburgh (2005), Petkova and Zhang (2005), Santos and Veronesi (2006), and Yogo (2006).

(e.g., Garleânu, Kogan, and Panageas (2012)). The assumption that only a few agents benefit from innovation is supported by evidence from patents, which is the main empirical proxy for innovation. For instance, Acemoglu, Akcigit, and Celik (2014) note that only a small fraction of all patents filed are highly cited, with the rest receiving almost no citations and creating almost no value for the patent owner.

Because of displacement risk, the majority of investors should tilt their portfolios to growth stocks "as a hedge against the potential wealth reallocation that may result from future technological innovation" (Kogan et al. (2017), p. 2). Within our context, investors have greater desire to invest in growth stocks when the correlation between their income growth and returns to growth stocks is especially low. These low correlations can signal that their income and the future potential of growth firms may be diverging, and thus they need to increase their holdings of growth stocks even further.

B. Data and Summary Statistics

In this section, we provide a brief description of our data sources and present summary statistics for our main variables.

B.1. Aggregate Income Risk and the HML Factor

We obtain HML returns from Ken French's data library, and quarterly U.S. income data from the Bureau of Economic Analysis (BEA).⁹ The sample period is from Q1 1960 to Q4 2011. The summary statistics reported at the top of Table I demonstrate that HML returns are positive on average and highly volatile. Specifically, the quarterly mean and standard deviation of the value factor are 1.247% and 6.192%, respectively.

B.2. State-Level Income Risk

In addition to the aggregate income hedging motive, we consider income hedging at the state level. We explore state-level income hedging because income shocks are not diversifiable. As a result, the income risk relevant for asset prices need not be the national income risk, but rather more disaggregated measures of risk (e.g., Constantinides and Duffie (1996)). Our choice is also motivated by evidence in Korniotis (2008) and Korniotis and Kumar (2013) that state-level income shocks are systematic. To compute the state-level IHD measure, we use state income data from the Bureau of Labor Statistics (BLS).

⁹ Income is the BEA's personal income measure, which is the sum of wages and salaries, proprietors' income, and personal current transfer receipts less contributions for government social insurance. Our measure of income excludes income from investments (i.e., interest and dividends).

Table I Summary Statistics

This table reports summary statistics of quarterly HML returns and state-level HML returns over the sample period. State-level HML returns are calculated as the return from a value-weighted portfolio of stocks headquartered in a given state that goes long (short) in the stocks of firms with a book-to-market ratio in the highest (lowest) within-state quartile. We consider states with at least 30 firms headquartered in the state during a given quarter. The sample period is from Q1 1960 to Q4 2011.

Variable	Mean	Median	Std. Dev.	10th pctile	90th pctile
HML Portfolio Return	1.247	0.590	6.192	-4.956	9.770
State HML Returns					
AR	-6.556	-6.556	10.437	-13.936	0.824
AZ	0.674	1.105	9.170	-9.837	10.877
CA	0.615	0.017	7.302	-8.651	9.227
CO	1.939	1.141	9.529	-7.563	12.829
СТ	0.926	0.108	6.678	-6.520	8.447
FL	2.226	1.395	6.660	-4.931	10.650
GA	1.910	1.343	7.592	-6.277	11.465
IA	-1.146	-1.218	7.301	-9.508	7.721
IL	-0.057	-0.788	4.821	-5.556	6.484
IN	0.317	-0.111	8.217	-7.675	8.962
KS	1.591	1.753	9.951	-7.754	13.175
KY	0.751	1.560	10.335	-10.820	13.755
LA	0.976	0.104	12.040	-10.032	9.353
MA	0.606	0.163	8.052	-9.429	11.567
MD	1.483	0.809	8.144	-8.762	9.977
MI	1.045	0.069	11.067	-8.490	11.720
MN	1.219	0.934	7.261	-6.302	8.337
МО	1.477	0.964	7.900	-7.319	11.589
NC	1.142	0.686	8.191	-6.899	9.335
NH	3.028	2.409	8.684	-6.924	15.099
NJ	0.975	-0.151	9.634	-7.969	9.274
NV	0.737	1.409	11.139	-16.182	12.815
NY	0.878	0.671	6.854	-5.800	7.251
OH	0.635	-0.223	6.297	-5.643	6.764
OK	1.858	1.835	10.898	-12.046	14.701
OR	1.133	0.761	11.034	-13.738	14.202
PA	1.241	0.330	6.709	-6.563	9.234
SC	0.760	0.238	8.832	-10.691	9.986
TN	2.531	1.374	10.361	-6.798	10.984
TX	1.240	1.096	5.570	-4.960	7.224
UT	0.814	1.046	12.136	-12.086	14.681
VA	0.824	0.481	9.739	-7.804	11.216
WA	-0.405	-1.041	9.766	-10.824	10.848
WI	1.457	0.953	7.903	-6.507	9.514

C. Aggregating Income Data

One potential limitation of our analysis is that we use aggregate nationaland state-level data to compute IHD. To be more consistent with our model, we would have liked to use disaggregated income data from household surveys. Unfortunately, no U.S. household-level data set has a time dimension long enough to allow estimation of an income hedging series that we could use in asset pricing tests. The use of aggregate income data is thus a necessary compromise.

Moreover, using aggregate income to compute IHD is consistent with the consumption and portfolio life-cycle literature. A common assumption in this literature is that household income growth is correlated with aggregate shocks and *not* idiosyncratic shocks. For example, Cocco, Gomes, and Maenhout (2005) "allow for correlation between innovations to excess stock returns and [household] labor income shocks through the aggregate component" of income growth (p. 495). Michaelides and Gomes (2005) make a similar assumption. Overall, using aggregate income is a reasonable approach in capturing the systematic component of individual households' income hedging.

D. Estimates of IHD

In this section, we examine whether the IHD measure exhibits significant variation over time. For the estimation, we assume that the only assets available to investors are four portfolios, namely, the market (RMRF), value-minus-growth (HML), small-minus-big (SMB), and winners-minus-losers (UMD) portfolios.¹⁰ These portfolios are likely to capture the investment opportunity set of common investors reasonably well, and they are also the portfolios used for constructing common asset pricing factors.

D.1. Estimation Method

In our estimation, we replicate the inference methodology of the Bayesian worker in the model (see equation (2)). Specifically, we first estimate the vector of conditional covariances $\widehat{\Sigma}_{y,r,t}$ between the returns of the four portfolios (RMRF, HML, SMB, UMD) and income in first-differences. To calculate the aggregate-level IHD, we use the covariances between the four portfolios and aggregate income in first-differences, while for the state-level IHD, we use the covariances between the four portfolios and state-level income in firstdifferences. Next, we estimate the conditional variance-covariance matrix for the returns of the four portfolios ($\widehat{\Sigma}_{r,t}$). Finally, we multiply the negative value of the vector of income covariances by the inverse of the covariance matrix of returns ($-\widehat{\Sigma}_{r,t}^{-1}\widehat{\Sigma}_{y,r,t}$). The output of this multiplication is a vector of scaled covariances. We focus on the element corresponding to the HML factor, which is the IHD estimate for the HML factor.

To capture time-variation in the IHD estimates, we use a rolling estimation approach. Specifically, we set the rolling estimation window to 10 years (40 quarters) and compute new covariance estimates each quarter. This procedure yields a time series for IHD covering the period from Q1 1970 to Q4 2011. In

 $^{^{10}}$ In Table IA.III, we construct the IHD measure using different assumptions regarding the sets of financial assets that are available to investors.

Table II, we present summary statistics for IHD. We use the 10-year window to be consistent with the recent macro-finance literature. For example, Marfè (2017) uses 10-year rolling windows and quarterly data to calculate a time series of variance ratios in asset pricing tests.¹¹

D.2. Time-Variation in IHD Estimates

The estimates of the HML hedging demand in Table II indicate that statelevel IHDs vary over time. For example, the IHD of California's income with aggregate HML returns ranges from $-3.100 (10^{\text{th}} \text{ percentile})$ to $6.623 (90^{\text{th}} \text{ percentile})$. Further, the standard deviation of the state-level IHD estimates is sizeable across states, varying from 2.248 (TN) to 18.218 (ND). The substantial time-variation in the state-level hedging measures is also clearly seen in Figure 1, Panel A. In the figure, we plot for each quarter the 10^{th} and 90^{th} percentile of the cross-sectional distribution of IHD. In almost all periods, some states exhibit positive and some exhibit negative IHD. Each quarter, we also compute the equal-weighted average IHD across states (State Avg), and report summary statistics for this measure at the bottom of Table II. The average IHD across states exhibits substantial time-series variation with a standard deviation of 4.654, and ranges from $-3.262 (10^{\text{th}} \text{ percentile})$ to 8.849 (90th percentile).

Last, we compute the IHD between HML returns and aggregate U.S. income in first-differences (Aggregate U.S. Income). We report summary statistics for this measure in Table III. Similar to the state-level results, we find strong time-series variation in the aggregate IHD. The strong time-variation in the national hedging term is also evident in Figure 1, Panel B.¹² In the figure, we plot the time series of the national IHD and the 95% confidence intervals of the average IHD. As shown in the figure, in many periods the national hedging term is outside these confidence interval bands.¹³

¹¹ We also consider alternative rolling window lengths for estimating IHD ranging from 8 to 15 years. We present these results in Table IA.II of the Internet Appendix. We find that lengthening the estimation window to 11, 12, or 13 years yields a statistically significant relation between IHD and HML returns. However, the main results fail to hold when extending the window to 14 or 15 years or shortening the window to 8 or 9 years. Among the alternative windows under consideration, the baseline 10-year window does not yield the strongest results statistically. Instead, the 12-year window yields the economically and statistically strongest IHD predictor (coefficient = -0.328, *t*-statistic = -2.77), which should alleviate concerns that our baseline window choice is the result of cherry-picking.

 12 Additionally, we present the time series of conditional correlations computed in the same way as the IHD measure in Figure IA.1 of the Internet Appendix.

 13 To further understand the time-series properties of the national IHD, we regress it on various combinations of lagged IHD, the set of factor premium predictors and macroeconomic measures from our baseline HML specifications, and an NBER recession indicator. We report the regression results in Table IA.IV of the Internet Appendix. We find that the IHD measure is highly persistent, with a very strong AR(1) coefficient across all specifications. Controlling for lagged IHD, few other explanatory variables are significantly related to IHD. In particular, of the factor premium predictors and baseline macroeconomic measures, IHD exhibits a strong positive relation only with unexpected GDP growth (*t*-statistic = 1.98).

Table II Summary Statistics: State-Level IHD Estimates

This table reports the state-level IHD estimates of quarterly state income growth and national HML returns over the sample period. The state-level IHD is based on conditional covariances between quarterly state income growth and national HML returns over the past 40 quarters for each quarter during the sample period with 40 trailing observations. We also report the equal-weighted average for all statistics across states (State Avg). The sample period is from Q1 1970 to Q4 2011.

			IHD Estimat	es	
State	Mean	Median	Std. Dev.	10th pctile	90th pctile
AK	-0.162	-0.543	5.379	-6.144	7.544
AL	-1.749	-1.354	2.486	-1.491	5.156
AR	0.224	-0.521	4.148	-5.489	4.437
AZ	-0.826	-0.524	3.019	-2.685	4.164
CA	-1.025	-0.219	3.750	-3.100	6.623
CO	-2.045	-2.121	4.506	-3.274	8.757
CT	-0.666	-0.941	3.901	-2.851	4.513
DC	-5.455	-3.133	14.536	-13.570	30.19
DE	-3.892	-2.296	5.180	-1.159	12.103
FL	-0.930	-0.321	2.535	-2.494	4.689
GA	-1.609	-0.820	3.380	-1.950	5.637
HI	0.326	0.832	4.187	-5.972	4.735
IA	-5.638	-3.015	7.136	-1.972	14.111
ID	-0.979	-0.710	5.102	-5.066	9.204
IL	-1.365	-0.260	3.948	-3.194	6.205
IN	-2.314	-2.332	2.780	-1.114	5.735
KS	-4.005	-3.915	3.254	1.022	8.066
KY	-0.878	-0.751	3.222	-2.997	5.119
LA	-2.309	-1.739	3.439	-1.863	7.454
MA	-1.412	-1.220	4.068	-4.142	6.89
MD	-1.208	-1.468	3.886	-4.407	6.50
ME	0.476	0.541	2.996	-4.274	4.139
MI	-3.746	-3.152	4.312	-0.859	9.899
MN	-2.801	-1.932	5.715	-3.281	11.037
MO	-2.114	-2.633	3.605	-3.495	6.733
MS	-0.458	-0.188	3.168	-3.665	5.032
MT	-7.714	-0.871	13.553	-2.229	31.110
NC	-1.330	-0.980	2.659	-0.523	5.038
ND	-8.017	-4.894	18.218	-6.586	41.711
NE	-6.668	-6.481	4.958	0.535	12.285
NH	-0.754	0.482	4.520	-4.075	8.418
NJ	-1.679	-1.766	3.864	-3.490	7.182
NM	0.322	1.261	3.485	-3.826	5.640
NV	-3.578	-2.575	4.531	-1.349	11.007
NY	-1.056	-2.177	7.747	-10.919	11.345
OH	-2.930	-2.780	2.650	-0.036	6.382
OK	-1.593	0.356	4.521	-2.291	7.560
OR	-0.790	-0.731	2.402	-2.018	4.472
PA	-1.381	-0.289	3.040	-1.912	6.035
RI	0.669	0.078	2.525	-3.837	2.049
SC	-1.581	-2.062	2.592	-1.658	4.423

(Continued)

			IHD Estimat	tes	
State	Mean	Median	Std. Dev.	10th pctile	90th pctile
SD	-5.004	-4.654	10.542	-7.539	21.358
TN	0.223	0.466	2.248	-2.541	3.247
TX	-1.287	-1.111	3.708	-3.715	6.999
UT	-0.608	-0.141	2.447	-2.336	4.453
VA	1.049	1.116	3.676	-5.065	4.021
VT	-1.788	-1.403	2.482	-0.466	5.814
WA	-5.611	-5.484	3.495	1.243	10.279
WI	-0.658	-0.057	3.048	-2.477	5.201
WV	-0.574	-1.373	5.025	-6.825	7.523
WY	-2.288	-0.594	5.790	-2.957	13.054
State Avg	-1.984	-1.400	4.654	-3.262	8.849
U.S. Income	-1.123	-0.115	3.269	-2.334	6.014

Table II—Continued

Taken together, our IHD estimates indicate that HML returns exhibit strong time-variation in their comovement with income at both the state and the aggregate level. The time-varying IHD estimates and the time-varying popularity of value/growth style portfolios (e.g., Kumar (2009)) suggest that switching between value and growth style portfolios might reflect the time-varying IHD for these styles.

III. Validating Hedging Demand Estimates

In this section, we present results from two validation tests, which are designed to investigate whether the IHD estimates for the HML factor capture actual time-variation in demands for value and growth stocks. First, we examine whether our covariance-based demand estimates are correlated with the portfolio weights of actual retail investors. Second, we investigate whether our IHD estimates are related to aggregate mutual fund flows.

A. Validation Using the Brokerage Data

In the first validation test, we use data on quarterly portfolio positions of retail investors at a major U.S. discount brokerage over the 1991 to 1996 period.¹⁴ To align these holdings with our panel of state-level HML demand estimates, we calculate the relative weight in value and growth stock portfolios at the end of each quarter by investors located in each U.S. state. In particular, at the end of each quarter, we aggregate the holdings of all investors located in a given state in value and growth stocks. We define value (growth) stocks as those stocks with book-to-market in the highest (lowest) within-state quartile. Next, we denote the total holdings in value (growth) stocks of state *s* investors

 $^{^{14}}$ See Barber and Odean (2001) for details about the brokerage data.

Table III

Summary Statistics: Income Hedging Demand (IHD) Estimates for the HML Factor and Factor Predictors

This panel reports summary statistics for the aggregate IHD measure, which is based on the conditional covariances between quarterly U.S. income growth and the returns on the national HML portfolio (equation (3)) over the sample period. We also report summary statistics for factor predictors, which include the dividend yield on the value-weighted CRSP market index over the previous 12 months (DIV), the yield on the three-month T-bill (YLD), the difference between the average yields of Treasury bonds maturing in more than 10 years and those maturing in three months (TERM), and the difference between the average yields of bonds with a Moody's rating of AAA and bonds with a Moody's rating of BAA (DEF). We also report summary statistics for various macroeconomic controls, including industrial production growth, quarterly inflation, the Conference Board's Leading Economic Indicator (LEI) Index, and unexpected quarterly U.S. GDP growth (residuals from an AR(1) model). Further, we report summary statistics for value factor predictors, including the HML RISK-RETURN measure, the book-to-market (BM) spread (log book-to-market ratio of value stocks minus log book-to-market ratio of growth stocks), small book-to-market spread (book-to-market spread calculated among stocks with market capitalization falling in the lowest tercile of NYSE market capitalization in a given quarter), the difference in dividend yield on value stocks and the dividend yield on growth stocks over the previous 12 months, and lagged one-month HML return. Finally, we report summary statistics for various state-level control variables, including the average and maximum market capitalization of firms headquartered in a given state, and the growth in Office of Federal Housing Oversight (OFHEO) state house price indices. The sample period is from Q1 1970 to Q4 2011.

Variable	Mean	Median	Std. Dev.	10th pctile	90th pctile
IHD Estimates for the HML Factor					
Baseline, National Income Risk	-1.123	-0.115	3.269	-2.334	6.014
Baseline, State Income Risk	-1.661	-0.462	2.908	-1.303	5.816
Value-Growth	-4.101	-4.984	9.551	-8.967	13.168
Factor Predictors					
Dividend Yield (DIV)	2.515	2.391	0.980	1.254	3.988
Yield on 3-month T-bill (YLD)	5.387	5.140	3.192	0.940	9.060
Term Spread (TERM)	1.681	1.825	1.322	-0.150	3.310
Default Spread (DEF)	1.117	0.990	0.467	0.670	1.740
Macroeconomic Controls					
Industrial Production Growth	0.554	0.736	1.709	-1.461	2.481
Inflation Rate	1.079	0.889	0.880	0.311	2.429
Conf Board LEI Index	83.846	80.250	21.279	57.300	112.900
Unexpected Quarterly GDP Growth	-0.260	-0.262	3.329	-4.244	3.423
Value Factor Predictors					
RISK-RETURN	9.846	8.786	4.680	4.893	16.838
Value Spread	1.123	1.094	0.169	0.939	1.341
Small Stock Value Spread	1.178	1.144	0.158	1.013	1.430
Value Div Yield – Growth Div Yield	2.078	2.200	1.367	0.061	3.810
Lagged 1m HML Return	0.298	0.450	2.641	-2.960	3.720
State-Level Controls					
Average Market Cap (\$ Millions)	1.094	0.521	1.260	150.176	2915.181
Maximum Market Cap (\$ Millions)	42.929	14.623	72.453	2.043	125.429
State Housing Growth	1.108	1.055	2.124	-1.032	3.367

as $V_s(G_s)$ and calculate the relative weight in value versus growth stocks as follows: $VG_s = \frac{V_s - G_s}{V_s + G_s}$. To eliminate the effect of passive changes in portfolio weights due to price changes, we set the price of each stock equal to its average price during the 1991 to 1996 sample period.

To compare these direct measures of relative value-minus-growth holdings with our theoretical demand estimates, we regress the actual state-level valueminus-growth portfolio weights on the state-level IHD of the HML factor. In the regressions, we include additional controls (e.g., dividend yield, term spread, industrial production growth, inflation rate) to account for factors besides income hedging that could generate the observed variation in state-level HML holdings.

Apart from the IHD measure, we include another theory-motivated control variable. In particular, following equation (5), we compute a risk-return term based on the return of the HML portfolio scaled by the inverse of the variance-covariance matrix of returns. This scaled return is the element of the vector $\widehat{\boldsymbol{\Sigma}}_{r,t}^{-1}\widehat{\boldsymbol{\mu}}_{r,t}$ that corresponds to the HML. We denote the scaled return by the RISK-RETURN term and compute it using the same rolling-window estimation approach as the IHD measure.

We present the results from within-state panel regressions in Panel A of Table IV. In all specifications, we find that the observed value-minus-growth portfolio weights covary positively with our IHD measure. These positive estimates are consistent with the theoretical predictions of our model. In particular, as the income hedging potential of the HML factor increases, so does the IHD. Therefore, investors demand more of the value-minus-growth asset, increasing their weight in the overall portfolio. This evidence suggests that our hedging demand estimates capture the actual behavior of retail investors at the state level reasonably well.

B. Validation Using Aggregate Mutual Fund Flows

In our second validation test, we examine whether our IHD estimates are related to aggregate mutual fund flows. Since many investors allocate a significant proportion of their financial wealth into mutual funds (e.g., Polkovnichenko (2005)), it is possible that our measure of income hedging captures at least part of the time-variation in investors' relative preferences for mutual fund styles. To investigate this possibility, we examine whether net flows into value and growth mutual funds are correlated with our IHD measure.

We use data on quarterly mutual fund returns and total net assets from the CRSP mutual fund database to calculate fund-level flows. Following Cooper, Gulen, and Rau (2005), value funds are those with the style identifier *value*/*val* in the NAME field and growth funds are those with the style identifier *growth*/*gr*/*grth*. Following Frazzini and Lamont (2008), we calculate net flows for each fund as the difference between end-of-quarter total net assets (TNA) and the fund's calculated counterfactual end-of-quarter TNA. Counterfactual

Table IV Validation Test Estimates Panel A reports coefficient estimates from within-state panel regressions of the relative value-growth portfolio weight on state-level IHD and a return, and a set of factor predictors, macroeconomic controls, and value factor predictors described in Table III. We calculate the mutual fund flows as in Frazzini and Lamont (2008) and classify value and growth funds as in Cooper, Gulen, and Rau (2005). The estimation period is from Q1 1985 to set of factor predictors, macroeconomic controls, and value factor predictors described in Table III. We calculate the relative value-growth portfolio all individuals in a given U.S. state using investor data from a large U.S. discount brokerage. The estimation period is from Q1 1991 to Q4 1996. B reports coefficient estimates from within-fund panel regressions of mutual fund flows on national-level IHD estimates, one-guarter-lagged fund weight as the difference between end-of-quarter value and growth holdings divided by the sum of end-of-quarter value and growth holdings across t-statistics computed using standard errors adjusted for clustering at the state level are reported in parentheses below coefficient estimates. Panel $Q4\,2011.t$ -statistics computed using standard errors adjusted for clustering at the fund level are reported in parentheses below coefficient estimates.

		Panel A: State	e-Level Value-Growth Por	tfolio Weight	
Regressor	(1)	(2)	(3)	(4)	(2)
IHD	0.023	0.026	0.025	0.027	0.028
	(2.34)	(2.46)	(2.42)	(2.66)	(2.55)
RISK-RETURN	0.044	0.061	0.057	0.043	0.054
	(2.58)	(2.26)	(2.16)	(2.24)	(2.01)
DIV		-1.872			-1.581
		(-1.14)			(-0.75)
ALD		-0.001			-0.007
		(-0.04)			(-0.19)
TERM		0.020			-0.006
		(0.69)			(-0.26)
DEF		-0.029			-0.105
		(-0.15)			(-0.86)
Ind Prod Growth			0.018		-0.008
			(00.00)		(-0.56)
Inflation Rate			0.089		-0.002
			(1.18)		(-0.05)
LEI Index			0.003		0.022
			(0.63)		(1.79)
Unexp GDP Growth			-0.001		-0.001
			(-0.09)		(-0.07)

(Continued)

Income Hedging, Dynamic Style Preferences, and Return Predictability 2075

		Table	IV—Continued			
			Panel A: State-Le	vel Value-Growth Por	tfolio Weight	
Regressor	I	(1)	(2)	(3)	(4)	(5)
Value Spread					0.630	-1.730
					(0.61)	(-1.38)
Small Value Spread					0.082	2.902
					(0.16)	(3.02)
Value DivYld – Growth DivYld					-0.088	-0.062
					(-1.92)	(-1.56)
Lagged 1m HML Return					0.008	0.011
					(2.76)	(3.51)
Adj. R^2	0	.076	0.092	0.081	0.099	0.108
Observations	-	717	717	717	717	717
			Panel B: Fund-Lev	el Mutual Fund Flows		
	All	Funds	Small	er Funds	Large	r Funds
	Value Funds	Growth Funds	Value Funds	Growth Funds	Value Funds	Growth Funds
Regressor	(1)	(2)	(3)	(4)	(5)	(9)
IHD	0.007	-0.005	0.005	-0.007	0.004	-0.003
	(3.65)	(-3.52)	(1.38)	(-2.88)	(2.79)	(-2.73)
RISK-RETURN	0.031	0.002	0.046	0.004	0.023	0.004
	(11.05)	(1.24)	(8.29)	(1.14)	(9.40)	(3.07)
One-qtr lagged fund return	0.133	0.364	-0.188	0.213	0.292	0.350
	(1.44)	(6.70)	(-1.21)	(2.15)	(3.67)	(7.81)
DIV	-1.177	-1.781	-1.051	-1.586	-0.792	-0.991
	(-3.23)	(-6.91)	(-1.64)	(-3.66)	(-2.54)	(-5.03)
ALD	0.029	0.032	0.087	0.077	0.006	0.010
	(4.10)	(6.08)	(6.09)	(8.55)	(1.09)	(2.53)
TERM	0.045	0.032	0.111	0.078	0.015	0.007
	(5.88)	(6.06)	(7.23)	(7.92)	(2.46)	(1.71)
						(Continued)

2076

The Journal of Finance®

		Table IV-	-Continued			
		I	anel B: Fund-Leve	با Mutual Fund Flow	x	
	All	Funds	Smalle	er Funds	Largei	Funds
Regressor	Value Funds (1)	Growth Funds (2)	Value Funds (3)	Growth Funds (4)	Value Funds (5)	Growth Funds (6)
DEF	-0.143	-0.091	-0.197	-0.115	-0.060	-0.035
Ind Prod Growth	(-0.001)	(-6.02)	(-4.94) -0.003	(-4.14) 0.004	(-3.00) -0.001	(16.2-)
Inflation Bate	(-0.28) -0.009	(4.08) 0.013	(-0.57) -0.022	(0.89) 0.007	(-0.20)	(5.19) 0.015
	(-1.29)	(2.94)	(-1.87)	(0.84)	(0.24)	(4.04)
LEI Index	-0.011	-0.011	-0.020	-0.018	-0.006	-0.006
	(-13.80)	(-19.32)	(-11.94)	(-16.42)	(-9.23)	(-12.71)
Unexp GDP Growth	0.001	0.003	-0.002	0.003	0.002	0.002
	(0.46)	(3.10)	(-0.95)	(1.75)	(1.50)	(2.86)
Value Spread	0.504	0.211	0.520	0.278	0.292	0.227
	(3.58)	(2.03)	(2.09)	(1.69)	(2.38)	(2.77)
Small Value Spread	-0.859	-0.621	-1.178	-0.903	-0.498	-0.386
Moline Dirveld County Dirveld	(-6.55)	(-6.35)	(-5.09)	(-5.88)	(-4.30)	(-4.97)
	(5.17)	(18.60)	(4.87)	(15.60)	(3.20)	(10.65)
Lagged 1m HML Return	0.005	-0.001	0.005	-0.004	0.004	-0.000
	(5.06)	(-1.63)	(2.93)	(-2.77)	(4.87)	(-0.89)
Adj. R^2	0.044	0.037	0.045	0.041	0.040	0.028
Observations	70,624	128,069	34,080	63, 273	36,544	64,796
Value-Growth	0	.012	0.	012	0.0	200
IHD loadings	(2)	5.08)	2)	.48)	(14	.80)

end-of-quarter TNA is the observed beginning-of-quarter TNA multiplied by the fund's observed return over the quarter plus a pro rata share (based on beginning-of-quarter TNA) of the total dollar flow to the mutual fund sector during the quarter.

To compare the actual net flows into value and growth mutual funds with our theoretical demand estimates, we regress fund-level net flows on the IHD of the HML factor. Panel B of Table IV presents estimates of such within-fund panel regressions, in which we also include a set of control variables. This set includes the RISK-RETURN term, the lagged fund returns to account for the performance-flow relation (e.g., Chevalier and Ellison (1997)), and macroeconomic variables such as industrial production growth and the inflation rate. See Section IV.A for a detailed description of all the control variables.

The regression estimates in column (1) show that average net flows into value funds covary positively with our IHD measure (estimate = 0.007, *t*-statistic = 3.65). In contrast, the regression estimates in column (2) show that average net flows into growth funds covary negatively with our IHD measure (estimate = -0.005, *t*-statistic = -3.52). Further, we find consistent results from specifications that focus on subsamples of smaller and larger funds (see columns (3) to (6)).

Taken together, these findings suggest that time-variation in the IHD estimate for HML reflects hedging-induced variation in investor demand for value and growth stocks. We conclude that income hedging appears to be an important determinant of style shifts among mutual fund investors.

IV. Income Hedging and Return Predictability

In this section, we test our key model prediction summarized in equation (5), which posits that an increase in the IHD of HML would lead to lower future HML returns. We test this key conjecture using a series of predictability regression models.

A. Return Predictability Regression Specifications

In our empirical analysis, we estimate the following regression specification:

$$r_{t+1}^{HML} = b_1 IHD_t^{HML} + \boldsymbol{b}' \boldsymbol{X}_t + \boldsymbol{u}_{t+1},$$

where r_{t+1}^{HML} is the one-period-ahead HML return, IHD_t^{HML} is the element of the vector $-\widehat{\Sigma}_{r,t}^{-1}\widehat{\Sigma}_{y,r,t}$ that corresponds to the HML, and **X** is a vector of control variables.

In our empirical regressions, we are mainly interested in the sign and statistical significance of the coefficient estimate b_1 . According to our model (see equation (5)), b_1 should be negative, so that a decrease in IHD, which corresponds to a decrease in the income hedging potential of HML, lowers current prices and increases future returns. We calculate IHD estimates for the HML factor based on information at the end of the previous quarter. We use this end-of-quarter demand to predict the return of the HML portfolio over the subsequent quarter and over each month of the subsequent quarter. Specifically, we estimate the IHD of HML with 10 years of quarterly income data as of the end of quarter q and predict the return of the HML portfolio during quarter q + 1 and during each of the three months of quarter q + 1. For example, on December 31, we estimate demands using data until December 31 and then predict the HML return over the quarter ending March 31 as well as the one-month HML returns for the months of January, February, and March.

We use this estimation approach with mixed timing because the income data are available only at the quarterly frequency and the theoretical demand estimates can be calculated only quarterly. Therefore, we consider one-quarterahead predictability regressions. However, we also examine the strength of return predictability within the quarter with one-, two-, and three-month-ahead forecasting regressions.

In all of the predictability regressions, we account for a set of factor predictors, macroeconomic controls, and HML predictors. Conditional on these control variables, an increase in the IHD measure should not reflect improving investment opportunities in the overall market. Instead, this increase is likely to be driven by shifts in the income hedging potential of the HML portfolio.

The choice of control variables is motivated by existing literature. The set of controls includes the dividend yield (DIV) of the market index (CRSP VW index) over the previous 12 months, which has been shown to be associated with slow mean reversion in stock returns over the previous 12 months (Keim and Stambaugh (1986), Campbell and Shiller (1988), Fama and French (1988)). We control for the yield on the three-month T-bill (YLD), which is negatively related to future stock market returns and proxies for expectations of future economic activity (Fama (1981), Fama and Schwert (1977a)).

We also control for the term spread (TERM), which is the difference between the average yield on Treasury bonds with 20 years to maturity and the average yield of T-bills maturing in three months. The term spread is closely related to short-term business cycles (Fama and French (1988)). We account for the default spread (DEF), which is the difference between the average yield of bonds rated BAA by Moody's and the average yield of AAA-rated bonds. The default spread tracks long-term business cycle conditions (Fama and French (1988)). Further, the set of control variables includes important macroeconomic predictors such as industrial production growth, the inflation rate, the Leading Economic Indicator (LEI) index, and the unexpected component of GDP growth based on an AR(1) process.

Last, we account for important HML predictors. Specifically, we control for the value spread and the small-stock value spread (Cohen, Polk, and Vuolteenaho (2003)), the spread between the dividend yield of value firms and the dividend yield of growth firms over the past 12 months, as well as the previous month's HML return (Wahal and Yavuz (2013)). We also include in the regressions the RISK-RETURN term to capture variation in the investment opportunity set of the HML portfolio.¹⁵

B. One-Quarter-Ahead Return Predictability Regressions

We present our baseline predictability results in Table V, Panel A. We present estimates from one-quarter-ahead regressions and from one-, two-, and threemonth-ahead regressions. In all specifications in Panel A, we calculate hedging demand estimates using national income (in first-differences). Our sample period is from Q1 1970 to Q4 2011.

We begin our analysis with one-quarter-ahead predictive regressions. In regression (1) of Panel A, we estimate a predictability regression using only the control variables. We find that the adjusted R^2 for this regression is 0.086. Most of the control variables are either statistically insignificant or marginally significant, with *t*-statistics ranging from -1.08 to 1.99. As expected, the two strongest predictors of the HML factor are the value spread and the difference in dividend yields between value and growth stocks (estimates = 19.293 and 1.343, *t*-statistics = 1.99 and 1.82, respectively).

In specification (2) of Panel A, the independent variable is the IHD associated with the HML factor. We find that this variable has a negative and statistically significant coefficient estimate (estimate = -0.295, *t*-statistic = -2.54). This estimate implies that a one-standard-deviation decrease in IHD for the HML factor is associated with a $0.295 \times 3.269 = 0.96\%$ increase in the HML return next quarter. Relative to the average quarterly HML returns over the sample period (=1.247\%), this magnitude is economically meaningful. Further, the adjusted R^2 for this regression increases to 0.094. Overall, the estimates in specification (2) suggest that hedging-induced demand has an economically meaningful effect on future HML returns.

C. Monthly Return Predictability Estimates

Next, we examine within-quarter predictability and estimate one-, two-, and three-month ahead regressions. We present the monthly forecasting results in columns (3) to (8) of Panel A. For each month, we estimate regressions with and without the IHD estimates. The results suggest that our IHD measure has the strongest effect on returns during the one-month-ahead period. In specification (4), the estimate on the HML demand is -0.181 and the *t*-statistic is -2.35. We also find that the predictive power of the IHD weakens during the next two months.

We also formally examine long-term predictability following the same methodology as the long-term predictability literature (e.g., Lettau and Ludvigson (2001a)). This literature suggests that to assess long-term effects, one

¹⁵ According to equation (5), the RISK-RETURN term is the element of $\widehat{\Sigma}_t^{-1} \widehat{\mu}_t$ that corresponds to the HML. We compute this term using the same 10-year rolling-window estimation approach as the IHD measure.

	. –	Predictive R	egression E	stimates for	r the Value I	actor		
Panel A reports coeffici a set of factor predictor portfolios (RMRF, SMF B reports coefficient es We calculate the IHD , HML, UMD). The factor state level and returns the fraction of national (4) (Value-Growth), we Growth), we calculate t bias-adjusted estimate.	ent estimates s. We calculat 3, HML, UMD) timates from r measure for th r predictors an on the four po on the four po i financial wea s estartaly ca the IHD measu s using the me	from regressions of the IHD measur). The factor predi regressions of one- and HML factor usi re as described in ortfolios (RMRF, S. alth held by state alth held by state ure between the vi- ure between the vi- stand of Amihud, 87) adjusted stand	f one-quarter-al e according to el ctors are as des inonth-ahead H ng income chan Table III. In spe MB, HML, UMI residents (calcu sis for value and alue factor and l Hurvich, and V	nead and <i>h</i> -mont quation (3) using erribed in Table IML returns on t ges at the natio erfication (3) (St. D), and then ave lated using IRS growth portfoli. U.S. income gro Vang (2009). Th eported in paren	h-ahead HML re g income changes III. The estimati the national-leve nal level and the ate-Level), we ca rage these measurage data on capital os, and then tak wth, instead of ir e estimation per theses below coe	turns on the nat s at the national on period is from 1 IHD measure a e returns of the f leulate the IHD ures in each qua gains and divide te the difference. toome changes. I riod is from Q1.	ional-level IHD level and return a Q1 1970 to Q4 and a set of factd four portfolios (1 measure using i rter, weighting (end income). In end income). In specification (in specification (1970 to Q4 2011) ss.	measure and us on the four 2011. Panel pr predictors. RMRF, SMB, RMRF, SMB, ncome at the aach state by specification 1 (5) (Income 6), we report 1. <i>t</i> -statistics
				Panel A: Predict	ive Regressions			
	Value Facto	r, 1-qtr ahead			Value Factor, h	-month ahead		
${\rm Predictor,\ Month\ }t$	(1)	(2)	h=1 (3)	h=1 (4)	h=2 (5)	h=2 (6)	$\substack{h=3\\(7)}$	h=3 (8)
IHD		-0.295						-0.031
RISK-RETURN		(-2.54) -0.018 (-0.19)		(-2.35) 0.018 (0.31)		(-1.30) 0.033 (0.62)		(-0.47) -0.067 (-1.19)
DIV	0.404 (0.28)	-0.077 (-0.05)	0.291 (0.36)	0.083 (0.10)	0.051 (0.09)	0.045 (0.07)	0.081 (0.10)	-0.163 (-0.20)
ALD	0.180	0.326 (1.08)	-0.108 (-0.68)	-0.028 (-0.17)	(0.86)	0.176	0.145 (0.98)	0.182 (1.14)
TERM	0.499	(1.21)	0.376	0.434	0.111 (0.46)	0.120	0.009	(0.25)
DEF	0.213	-0.188 (-0.17)	0.054	-0.076 (-0.10)	-0.290 (-0.49)	-0.240 (-0.37)	0.378	0.080
Ind Prod Growth	(0.91)	0.292 (0.79)	(0.87)	0.243 (0.86)	0.111 (0.52)	0.133 (0.57)	-0.028 (-0.21)	-0.075 (-0.54)
								(Continued)

Table V

Income Hedging, Dynamic Style Preferences, and Return Predictability 2081

			Pa	nel A: Predictiv	re Regressions			
	Value Fa	ctor, 1-qtr ahead		1	Value Factor, <i>h</i>	<i>i</i> -month ahead		
$\mathbf{Predictor, Month} \ t$	(1)	(2)	h=1 (3)	h=1 (4)	h=2 (5)	h=2 (6)	h=3 (7)	h=3 (8)
Inflation Rate	0.698	0.366	0.928	0.782	-0.109	-0.117	-0.154	-0.317
	(0.76)	(0.40)	(1.90)	(1.66)	(-0.25)	(-0.25)	(-0.46)	(-0.88)
LEI Index	0.045	0.051	0.027	0.033	0.007	0.011	0.009	0.007
Unexp GDP Growth	(0.78) 0.286	(0.97) 0.279	(0.98) 0.182	(1.21) 0.176	(0.31) 0.020	(0.46) 0.015	(0.30) 0.085	(0.23) 0.089
4	(1.49)	(1.51)	(1.39)	(1.38)	(0.28)	(0.22)	(1.27)	(1.34)
Value Spread	19.293	20.071	7.922	9.079	9.951	11.012	1.248	-0.166
	(1.99)	(2.12)	(1.72)	(1.91)	(2.06)	(2.33)	(0.27)	(-0.04)
Small Value Spread	-7.484	-8.397	-3.619	-4.615	-4.943	-5.725	1.132	1.995
	(-1.08)	(-1.32)	(-1.03)	(-1.34)	(-1.64)	(-1.96)	(0.36)	(0.64)
Value DivYld – Growth DivYld	1.343 (1.82)	1.596	0.901	1.012 (2.07)	0.396 (1 18)	0.401	0.001	(0.37)
1m HML Return	0.314	0.298	0.012	-0.001	0.336	0.328	-0.030	-0.025
	(1.35)	(1.29)	(0.00)	(-0.01)	(1.87)	(1.82)	(-0.22)	(-0.19)
Constant	-22.211	-21.545	-10.169	-10.663	-7.343	-8.333	-4.444	-2.386
	(-1.66)	(-1.63)	(-1.58)	(-1.55)	(-1.05)	(-1.18)	(-0.66)	(-0.36)
Adj. R^2	0.086	0.094	0.085	0.098	0.115	0.110	-0.038	-0.042
Quarters	168	168	168	168	168	168	168	168
		Panel	B: Predictive Regr	essions, 1-mon	th-ahead Valu	e Factor		
Variations in IHD Measure		National	State-Level	Nation Value–Gr	al owth	National	Na	tional
						Income Growth	Bias-	Adjusted
Sample Period1970-Predictor, Month t (1)	-2011	1970-2011 (2)	1970-2011 (3)	1970-2((4)	111	1970-2011 (5)	197)-2011 (6)
IHD		-0.181 (-2.35)	-0.184 (-2.11)	-0.06 (-2.16	57 5)	-0.250 (-2.30)		0.170 2.21)
							(Co	ntinued)

Table V—Continued

2082

The Journal of $Finance^{\mathbb{R}}$

g	l
1	l
n	l
<i>.</i> ;;	l
tt	l
2	l
2	l
Ŷ	l
	l
5	l
	l
Ð	l
	l
2	l
_ದ	l
H	l
-	L

		Pane	l B: Predictive Regre	ssions, 1-month-ahead	Value Factor	
Variations in IHD Measure		National	State-Level	National Value–Growth	National	National
Sample Period	1970–2011	1970–2011	1970–2011	1970–2011	Income Growth 1970–2011	Bias-Adjusted 1970–2011
Predictor, Month t	(1)	(2)	(3)	(4)	(2)	(9)
RISK-RETURN		0.018	0.026	0.040	0.024	0.028
		(0.31)	(0.44)	(0.66)	(0.40)	(0.49)
DIV	0.291	0.083	0.194	0.078	0.223	0.028
	(0.36)	(0.10)	(0.23)	(0.09)	(0.26)	(0.03)
YLD	-0.108	-0.028	-0.060	-0.018	-0.046	-0.029
	(-0.68)	(-0.17)	(-0.36)	(-0.11)	(-0.28)	(-0.17)
TERM	0.376	0.434	0.407	0.471	0.428	0.435
	(1.26)	(1.43)	(1.34)	(1.51)	(1.41)	(1.43)
DEF	0.054	-0.076	-0.088	0.183	-0.018	-0.121
	(0.07)	(-0.10)	(-0.11)	(0.25)	(-0.02)	(0.16)
Ind Prod Growth	0.234	0.243	0.260	0.248	0.253	0.246
	(0.87)	(0.86)	(0.91)	(0.88)	(0.90)	(0.87)
Inflation Rate	0.928	0.782	0.817	0.856	0.810	0.782
	(1.90)	(1.66)	(1.72)	(1.77)	(1.69)	(1.66)
LEI Index	0.027	0.033	0.035	0.038	0.039	0.036
	(0.98)	(1.21)	(1.27)	(1.36)	(1.41)	(1.34)
Unexp GDP Growth	0.182	0.176	0.174	0.184	0.175	0.176
Value Spread	7.922	9.079	8.723	9.863	8.986	8.836
	(1.72)	(1.91)	(1.83)	(2.14)	(1.90)	(1.86)
Small Value Spread	-3.619	-4.615	-3.876	-4.788	-4.505	-4.791
	(-1.03)	(-1.34)	(-1.13)	(-1.40)	(-1.30)	(-1.39)
Value DivYld –	0.901	1.012	0.993	0.855	1.017	1.007
Growth DivYld	(2.05)	(2.07)	(2.04)	(1.83)	(2.08)	(2.06)
1m HML Return	0.012	-0.001	-0.006	0.017	0.000	-0.001
	(0.09)	(-0.01)	(-0.05)	(0.14)	(0.00)	(-0.01)
Constant	-10.169	-10.663	-11.328	-12.050	-11.680	-10.663
ţ	(-1.58)	(-1.55)	(-1.64)	(-1.81)	(-1.70)	(-1.55)
Adj. R^2	0.086	0.094	0.085	0.098	0.115	0.110
Quarters	168	168	168	168	168	168

should focus on cumulative returns. We therefore compute cumulative HML returns over holding periods ranging from 1 to 24 months. We plot the estimates of the various long-term regressions in Figure IA.2 of the Internet Appendix together with ± 2 standard error bands. The standard errors use a dynamic Newey and West (1987) adjustment. We find that the IHD measure is a statistically significant predictor of cumulative HML returns over holding periods of up to 15 months. This finding suggests that the IHD-induced predictability is strong in the medium term.

We use the one-quarter-ahead regression estimates to examine whether the observed HML returns align with the HML returns implied by our model. Specifically, Figure 1 suggests that the value premium should be lower or negative in the second half of the 1980s and in the period 1992 to 2001, that is, during the tech boom. To examine whether these observations align with our model, in Figure IA.3 of the Internet Appendix we plot the expected HML returns implied by our main specification against the realized HML returns over the sample period. The correlation between the two series is 0.417 and is highly statistically significant, with a *p*-value of less than 0.0001. Further, we find that both the implied and realized HML returns are relatively low in the late 1980s and from 1992 to 2001.

To summarize, the results in Panel A of Table V indicate that our IHD for the HML portfolio can forecast future HML returns. This predictability is the strongest one month ahead but it is persistent for more than one year. In light of these findings, in the rest of our analysis, we focus exclusively on one-monthahead predictive regressions and provide additional evidence supporting the income hedging mechanism.

D. Robustness of Monthly Return Predictability Estimates

In Panel B of Table V, we focus exclusively on one-month-ahead predictability regressions and provide evidence from various robustness tests. To facilitate comparison, regression (1) in Panel B reports estimates from the predictability regression that uses only the set of control variables as predictors. Regression (2) presents the baseline results for the IHD measure based on national income.

In our first robustness check, we adopt a more disaggregated approach in computing the IHD. In this case, we first compute IHD at the state level. We then aggregate the state-level demands to the national level, weighting each state by the fraction of national financial wealth held by state residents.¹⁶ We report the estimation results in column (3) of Panel B. The coefficient estimate on the new IHD is almost identical to the estimate in column (2), although its statistical significance is weaker (*t*-statistic = -2.11).

In our next tests, we show that the significance of the IHD estimated with national income data is robust to alternative estimation methodologies. Specifically, in column (4), we separately calculate the IHD of the value and growth

¹⁶ The fraction of national financial wealth held by state residents is calculated using IRS data on capital gains and dividend income reported on tax returns at the state level.

portfolios with aggregate income and then take the difference. In column (5), we calculate the IHD using income growth instead of income in first-differences. In our baseline analysis, we use income in first-differences to be consistent with our model in Section I. The results in columns (4) and (5) indicate that the coefficient estimate on IHD remains negative and statistically significant.

Following Stambaugh (1999), we examine the potential biases in estimates of the HML demand coefficient. Stambaugh (1999) shows that the estimates in predictability regressions can be biased if the predictors are persistent and endogenous. As shown in Table IA.IV of the Internet Appendix, the IHD demand is persistent and thus it is possible that our original OLS estimates are biased. Our bias-adjustment estimation follows the multivariate method of Amihud, Hurvich, and Wang (2009). The results for the bias-corrected IHD are in column (6) of Table V, Panel B.

The corrected estimate in column (6) is -0.170 (*t*-statistic = -2.21). This value is very close to the baseline estimate in column (2) of Panel B (estimate = 0.181, *t*-statistic = 2.35). The two estimates are similar because the estimation bias is small. Specifically, we find that, even though the persistence in IHD is high (see the high AR(1) coefficient in Table IA.IV of the Internet Appendix), the level of endogeneity (i.e., the correlation between the error terms from the predictability regression and the AR(1) model of the predictor) is low (correlation = 0.05).

We also examine the robustness of our results to excluding episodes in which IHD exhibits sharp increases and declines, as these are potentially influential observations. In particular, we focus on two episodes: the sharp increase and subsequent decline in the early 1990s and the sharp permanent increase in Q4 2003. To remove the effect of these two episodes, we respectively exclude observations in the Q4 1990 to Q4 1992 and Q2 2003 to Q2 2004 periods. We also consider the case in which we exclude both of these periods. We report these results in Table IA.V of the Internet Appendix. In all cases, we find that our results are robust to excluding potentially influential observations. Interestingly, we find that our results are stronger, both economically and statistically, when we exclude the short-lived spike in IHD in the early 1990s.

In our final robustness test, we decompose IHD into its components. Our goal is to show that the variation in IHD that forecasts returns is related to the numerator (i.e., the income-HML covariance) and not the denominator (i.e., the HML variance). Specifically, we decompose the IHD measure into three components: (i) the income-HML covariance, (ii) the HML variance, and (iii) a residual term equal to IHD minus the sum of the income-HML covariance and negative one times the HML variance. We find that the income-HML covariance predicts a higher next-month HML return, with a *t*-statistic of 1.72. In contrast, the denominator and residual components have insignificant coefficient estimates, with *t*-statistics of just 0.83 and -0.83, respectively.

Overall, our results suggest that time-variation in the value factor is driven in part by the time-varying effectiveness of the HML factor as a hedge against income risk. In particular, our findings imply that a rise in the income hedging potential of the HML factor at the end of the quarter creates price pressure for HML-component stocks, resulting in lower future returns in the short to medium term.

V. Income Hedging and State-Level Asset Prices

Our analysis so far focuses on predicting the aggregate HML factor return. In this section, we examine whether state-level HML returns can be predicted more effectively using local measures of hedging-induced demand. This analysis is motivated by the observation that income risk faced by individuals is likely to vary geographically. Further, high levels of local stock ownership may be explained in part by income hedging motives. Van Nieuwerburgh and Veld-kamp (2009) show that if local investors have an initial information advantage on local stocks, they may specialize in these stocks and potentially hold a portfolio of local stocks that minimizes their income risk. Hence, local investors may optimally choose which local stocks to hold based on their hedging potential.

Our state-level analysis is also motivated by the local bias literature, which demonstrates that investors exhibit a preference for local stocks and these preferences could affect stock prices (e.g., Coval and Moskowitz (2001), Korniotis and Kumar (2013)). Specifically, Korniotis and Kumar (2013) find that withinstate ownership of state-headquartered firms is high among both retail and institutional investors, and that time variation in state-level ownership affects the returns of local firms.

In light of these findings, we examine whether hedging-induced shifts in local asset demands affect the returns of local value and growth portfolios. Specifically, we posit that if investors in a state overweight stocks of firms located in that state, then the local HML portfolio will exhibit low future returns when the local IHD is high. We examine this conjecture using state-level IHD estimates and the returns of the state-level HML portfolios.

A. State-Level HML Returns

We construct state-level HML portfolios using monthly stock returns, stock prices, and shares outstanding from CRSP. We restrict the sample to common stocks (i.e., share codes of 10 or 11). We merge these observations with firm headquarters data from Compustat. We also compute book-to-market ratios for each firm using data from Compustat. The book-to-market ratio is the sum of year-end book equity and balance sheet deferred taxes divided by year-end market equity. We assume the standard six-month lag between measurement and observation periods.

We calculate quarterly state-level HML portfolio returns by sorting stocks headquartered within each state using their book-to-market ratio. We then calculate the state HML return as the return of a value-weighted portfolio of stocks that assumes a long (short) position in the stocks of firms with a book-tomarket ratio in the highest (lowest) within-state quartile. To ensure that the state-level HML returns are economically meaningful, we limit attention to states with at least 30 firms headquartered in the state during a given quarter. Our portfolio construction procedure yields state HML time series for 35 states. We present summary statistics for state-level HML returns in Table I. The summary statistics indicate that state-level value factors offer positive average returns for almost all states, except AR, IA, IL, and WA. Further, state value factor returns exhibit high volatility. The standard deviation of returns ranges from 4.821% (IL) to 12.136% (UT).

B. Local Hedging Demand and Local Predictability

We examine the predictability of state-level HML portfolios using panel predictive regressions. In columns (1) to (3) of Table VI, we report estimates from within-state panel regressions of one-month-ahead state-level HML returns on state-level IHD. As described in Section II.D.1, state-level IHD is constructed using the covariances between state-level income and the returns of the four aggregate portfolios (RMRF, HML, SMB, UMD).¹⁷

The panel regression specifications contain the aggregate control variables used previously (see Tables IV and V) as well as state-level predictors. The state-level predictors include average firm size, maximum firm size, and average housing growth. The specifications also include state-level fixed effects. The estimation period for the panel regressions is from Q1 1970 to Q4 2011. We consider only 35 states in our estimation. This set includes states for which we have at least 30 firms so that state-level value and growth portfolios are well defined.

The estimates in Table VI show that state-level IHDs predict one-monthahead local HML returns. In the univariate panel regression (1), the coefficient estimate on the state-level IHD is -0.093 (*t*-statistic = -3.44). This estimate remains negative and statistically significant (estimate = -0.105, *t*-statistic = -3.53) even when we add all of the market-wide and state-level predictors (see columns (2) and (3)). Taken together, the state-level panel regressions provide evidence that growing local demand for the HML portfolio at the beginning of a month induces price pressure that generates a low return for the local HML portfolio during the following month.

C. Subsamples Based on Local Ownership

If local hedging demands generate predictable patterns in local HML returns, the evidence of predictability should be stronger or even exclusively concentrated among stocks with high local ownership. To test this conjecture, we examine the effects of local ownership on local HML return predictability.

We identify the percentage ownership of a firm that is local using retail investor data from a large U.S. discount brokerage house, which covers the

 $^{^{17}}$ For robustness, in Section IV of the Internet Appendix we construct the state-level IHD using state-level income and state-level HML portfolio returns. The predictability results for this alternative state-level IHD measure are very similar to those presented here.

	Estimat
	Regression
Table VI	Predictive
	tate-Level Panel
	Ś

ŝ

This table reports estimates from within-state panel regressions of the one-month-ahead state-level HML return on state-level IHD measures. The state-level HML return is calculated as the return on a value-weighted portfolio of stocks headquartered in a given state that goes long (short) in the stocks of firms with a book-to-market ratio in the highest (lowest) within-state quartile. The state-level IHD measure is constructed based on equation (3) using state-level income and returns on the four aggregate portfolios (RMRF, SMB, HML, UMD). The local ownership conditional state-level value factor is calculated by first conditioning on high- or low-local-ownership stocks (above or below median in the cross-section of CRSP stocks in a given month), and then calculating the return from a value-weighted portfolio of stocks headquartered in a given state that goes long (short) in the stocks of firms with a book-to-market ratio in the highest (lowest) within-state quartile. All specifications include factor predictors described in Table III. The estimation period is from Q1 1970 to Q4 2011. t-statistics computed using standard errors adjusted for clustering at the state level are reported in parentheses below coefficient estimates.

				High-Lo	cal-Ownership	Stocks	Low-Loo	cal-Ownership	Stocks
${\it Predictor, Month} \ t$	(1)	(2)	(3)	(4)	(2)	(9)	(2)	(8)	(6)
IHD	-0.093	-0.108	-0.105	-0.052	-0.100	-0.081	-0.072	-0.072	-0.096
	(-3.44)	(-3.78)	(-3.53)	(-1.83)	(-2.81)	(-2.07)	(-1.60)	(-1.37)	(-1.51)
RISK-RETURN	-0.068	-0.068	-0.071	-0.044	-0.071	-0.080	-0.059	-0.038	-0.038
	(-2.90)	(-2.45)	(-2.59)	(-1.74)	(-1.96)	(-2.22)	(-2.43)	(-0.90)	(-0.80)
DIV		0.425	1.172		-0.913	-0.554		0.794	1.006
		(0.92)	(2.37)		(-1.77)	(-0.91)		(1.13)	(1.22)
YLD		-0.453	-0.457		-0.237	-0.228		-0.409	-0.433
		(-4.92)	(-4.48)		(-1.93)	(-2.06)		(-2.54)	(-2.48)
TERM		-0.220	0.012		-0.269	-0.111		-0.149	-0.072
		(-1.36)	(0.06)		(-1.21)	(-0.46)		(-0.77)	(-0.30)
DEF		1.196	1.320		1.822	1.826		0.968	1.238
		(2.46)	(2.47)		(3.67)	(3.96)		(1.47)	(1.62)
Ind Prod Growth		-0.282	0.036		0.196	0.447		-0.161	0.096
		(-2.44)	(0.32)		(1.10)	(2.48)		(-0.93)	(0.52)
Inflation Rate		0.424	0.444		-0.234	-0.257		0.317	0.319
		(1.98)	(1.96)		(-0.54)	(-0.60)		(0.89)	(0.87)
LEI Index		0.025	0.070		-0.015	-0.005		0.024	0.056
		(1.76)	(3.47)		(-0.84)	(-0.20)		(1.41)	(1.89)

(Continued)

The Journal of Finance®

			Table V	I—Continue	d				
				High-L	ocal-Ownersh	ip Stocks	Low-Lo	ocal-Ownershi	p Stocks
Predictor, Month t	(1)	(2)	(3)	(4)	(2)	(9)	(2)	(8)	(6)
Unexp GDP Growth		0.387	0.287		0.255	0.137		0.247	0.177
		(8.88)	(6.58)		(2.77)	(1.73)		(3.61)	(2.32)
Value Spread		8.612	8.156		3.123	0.879		10.461	10.480
		(3.69)	(3.52)		(1.19)	(0.40)		(2.29)	(2.08)
Small Value Spread		-0.462	1.346		3.116	5.716		-3.105	-2.497
		(-0.30)	(0.72)		(1.19)	(2.21)		(-0.88)	(-0.63)
Value DivYld –		1.516	1.546		1.166	1.170		1.103	1.066
Growth DivYld		(9.11)	(8.40)		(5.73)	(5.53)		(4.87)	(4.54)
1m HML Return		-0.004	-0.002		-0.049	-0.040		0.068	0.049
		(-0.13)	(-0.06)		(-1.11)	(-0.88)		(2.42)	(1.53)
State Avg. Firm Size			-0.566			-0.181			-0.669
			(-1.95)			(-0.64)			(-2.38)
State Max Firm Size			0.010			0.010			0.004
			(3.12)			(2.80)			(1.28)
State Housing Growth			0.067			0.161			0.025
			(1.64)			(2.19)			(0.29)
State Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	\mathbf{Yes}	Yes	Yes
Adj. R^2	0.004	0.050	0.047	0.001	0.037	0.040	0.003	0.036	0.033
U.S. States	35	35	35	27	27	27	19	19	19
State-Quarter Observations	4,485	4,485	4,130	2,651	2,651	2,570	1,900	1,900	1,765

Income Hedging, Dynamic Style Preferences, and Return Predictability 2089

period from 1991 to 1996.¹⁸ For each month in the retail data set and for each firm, we measure the percentage of shares outstanding held by investors in the state where the firm is headquartered. Based on the average value of the monthly ownership measures for each firm, we sort firms into high or low local ownership categories.

For each U.S. state, we calculate the HML returns separately for high local ownership and low local ownership firms. For each local ownership category, we calculate state-level HML portfolio returns as before. Specifically, we sort all stocks headquartered in a given state and in a given local ownership category by book-to-market ratio. We then form a long(short) portfolio of stocks by valueweighting firms with a book-to-market ratio in the highest (lowest) withinstate quartile. To have meaningful ownership-based state HML portfolios, we continue to impose the 30-firm minimum requirement for each local ownership subsample when sorting on book-to-market ratio at the state level.

We present the predictability regression estimates for both local ownership subsamples in columns (4) to (9) of Table VI. For each local ownership-based subsample, we regress one-month-ahead state-level HML returns on the state-level IHD measures. The estimation results reveal that our evidence of predictability of the local HML portfolios by the local IHD is concentrated among firms with high local ownership. In particular, the coefficient estimate for the high local ownership subsample in column (6), where we control for all predictors, is -0.081 and statistically significant (*t*-statistic = -2.07). In contrast, the coefficient estimate for the low local ownership subsample in column (9) is weak (estimate = -0.096, *t*-statistic = -1.51).

D. Long-Short Portfolio Performance Estimates

The evidence from the state-by-state predictability regressions illustrates that local HML portfolio returns are related to the variability in local IHDs. Next, we construct a long-short portfolio that exploits the cross-sectional variation in state-level HML demand estimates. The performance of this portfolio provides out-of-sample evidence of the economic significance of our predictability regressions.

We define portfolios by sorting the state-level HML portfolios using our estimates for the state-level IHD measure. We form six portfolios. The Long (Short) portfolio is an equal-weighted portfolio of the three states with the lowest (highest) HML IHD at the end of a month and thus predicted to have the highest (lowest) HML return in the next month. The Long–Short portfolio captures the difference in the returns of the Long and Short portfolios. The final three portfolios, that is, portfolios 2 to 4, are equal-weighted portfolios of the remaining states sorted into terciles based on their respective HML demands. Portfolios are formed at the beginning of each quarter and are held for one month. The

 18 The retail sample is well represented across all U.S. states. In particular, the distribution of households across states is very similar between the retail investor sample and the Census data. See Barber and Odean (2000) and Appendix D of Korniotis and Kumar (2011) for additional details.

Table VII State-Level IHD Portfolios: Performance Estimates

This table reports performance estimates of a long-short portfolio defined using the IHD measure of the local value-growth factor. Component returns are those of value-weighted state-level value-growth portfolios. We report the performance of six portfolios: the Short portfolio, which is an equal-weighted portfolio of the three states predicted to have the lowest value-growth returns in the next month, the Long portfolio, which is an equal-weighted portfolio of the three states predicted to have the highest value-growth returns in the next month, the Long-Short portfolio, which captures the difference in the returns of the Long and Short portfolios, and portfolios 2, 3, and 4, which are equal-weighted portfolios of the remaining states sorted into terciles based on predicted value-growth returns in the next month. Portfolios are formed at the beginning of each quarter and are held for one month. Long-Short portfolio profits are then invested in the risk-free asset for the second and third months of each quarter. In Panel A, we report raw and characteristic-adjusted portfolio returns. Characteristic-adjusted returns are computed using the method of Daniel et al. (1997). *t*-statistics computed using Newey and West (1987) adjusted standard errors are reported in parentheses below the estimates. In Panel B, we report the standard deviation and Sharpe ratio for each portfolio. The portfolio formation period is from Q1 1970 to Q4 2011.

	Panel A: Po	ortfolio Returns
Portfolio	Raw Return	Char-Adj Return
1 (Short)	0.944	-0.107
	(2.18)	(-0.35)
2	0.641	0.222
	(1.53)	(0.82)
3	1.356	0.684
	(3.66)	(2.50)
4	1.257	0.465
	(3.85)	(2.46)
5 (Long)	2.312	1.419
	(5.02)	(3.48)
Long-Short	1.369	1.525
-	(2.73)	(2.77)
Quarters	168	168

Panel B: Portfolio Performance Characteristics (Excess raw returns)

Portfolio	Std Dev	Sharpe Ratio
1 (Short)	6.851	-0.055
2	5.705	-0.119
3	5.465	0.006
4	5.243	-0.012
5 (Long)	6.417	0.154
Long-Short	6.378	0.215

Long–Short portfolio profits are then invested in the risk-free asset for the second and third months of each quarter.

Table VII reports quarterly performance estimates of the six portfolios. In Panel A, we report raw and characteristic-adjusted portfolio returns. The characteristic-adjusted returns are computed using the Daniel et al. (1997) method. The *t*-statistics, which are computed using Newey and West (1987)

standard errors, are reported in parentheses below the estimates. In Panel B, we report the standard deviation and the Sharpe ratio for each portfolio. The sample period is from Q1 1970 to Q4 2011.

The performance estimates show that the Long portfolio significantly outperforms the Short portfolio. For example, the quarterly characteristic-adjusted return for the Long portfolio is 1.419% (*t*-statistic = 3.48), the Short portfolio has a characteristic-adjusted return of -0.107% (*t*-statistic = -0.35), and the Long–Short portfolio earns 1.525% per quarter (*t*-statistic = 2.77). We find similar results when we examine the portfolio Sharpe ratios. The Sharpe ratios of the Short, Long, and Long–Short portfolios are -0.055, 0.154, and 0.215, respectively.

We further examine the performance of our portfolios using various factor models. In particular, we estimate factor regressions for the Short, Long, and Long–Short portfolio returns. The factor models contain some combination of the following factors: the market excess return (RMRF), the size factor (SMB), the value factor (HML), the momentum factor (UMD), two reversal factors (short-term reversal (STR) and long-term reversal (LTR)), and the liquidity factor (LIQ). We report the quarterly alpha estimates and factor exposures in Table VIII. The *t*-statistics computed using Newey and West (1987) adjusted standard errors are reported in parentheses below the estimates.

Consistent with the performance estimates in Table VII, the alpha estimates of the Long–Short portfolio in Table VIII are positive and consistently significant. For example, the alpha estimate for the Long–Short portfolio from the model that includes all of the factors (RMRF, SMB, HML, UMD, STR, LTR, and LIQ; see column (12)) is 1.304% (*t*-statistic = 2.38). Overall, these performance estimates indicate that the predictive power of the HML demand estimates is strong out of sample, even when we consider various factor models to account for risk differences across portfolios.

VI. Additional Evidence and Robustness Tests

In this section, we present additional evidence on the relation between income hedging and return predictability that we document. We also relate our work to existing research and present various additional robustness tests.

A. Occupational Risk

We provide more evidence that our results are related to income hedging following Davis and Willen (2000). They show that across 10 occupation categories, electrical engineers face the highest income risk and auto mechanics the lowest. Similar to Davis and Willen (2000), we use data from the Current Population Survey (CPS) to construct annual income growth series for electrical engineers and auto mechanics and calculate occupation-specific IHD estimates. Since the CPS income growth series are only available from 1968

	Estimates
	Model
III	Factor
Table V.	Portfolios:
	IHD
	State-Level

(UMD), two reversal factors (short-term reversal (STR) and long-term reversal (LTR)), and the liquidity factor (LIQ). t-statistics computed using are excess returns of value-weighted state-level value-growth portfolios. We consider estimates of i) the Long portfolio, which is an equal-weighted portfolio of the three states predicted to have the highest value-growth returns in the next month, ii) the Short portfolio, which is an equal-weighted portfolio of the three states predicted to have the lowest value-growth returns in the next month, and iii) the Long-Short (L-S) portfolio, which month. Long-Short portfolio profits are then invested in the risk-free asset for the second and third months of each quarter. The factor models contain some combination of the following factors: the market excess return (RMRF), the size factor (SMB), the value factor (HML), the momentum factor This table reports factor model estimates of a long-short portfolio defined using the IHD measure of the local value-growth factor. Component returns captures the difference in the returns of the Long and Short portfolios. Portfolios are formed at the beginning of each quarter and are held for one Newey and West (1987) adjusted standard errors are reported in parentheses below the estimates. The estimation period is from Q1 1970 to Q4 2011.

				•					•		•	,
Factor	Long (1)	Short (2)	L_S (3)	Long (4)	Short (5)	L_S (6)	Long (7)	Short (8)	L_S (9)	Long (10)	Short (11)	L-S (12)
Alnha	1 030	0.215	1 345	0155	1 133	1 988	0 590	787 0	1 316	0.419	0.809	1 304
pudat.	(2.09)	(-0.78)	(2.59)	(0.37)	(-2.70)	(2.63)	(1.12)	(-1.26)	(2.41)	(0.91)	(-1.43)	(2.38)
RMRF	-0.026	-0.041	0.015	-0.008	-0.029	0.021	-0.057	-0.040	-0.017	0.010	0.016	-0.006
	(-0.35)	(-0.51)	(0.20)	(-0.12)	(-0.37)	(0.22)	(-0.79)	(-0.48)	(-0.17)	(0.14)	(0.18)	(-0.06)
SMB				0.383	0.372	0.011	0.292	0.269	0.023	0.326	0.299	0.027
				(3.97)	(3.47)	(0.10)	(3.62)	(2.45)	(0.20)	(3.95)	(2.75)	(0.23)
HML				0.530	0.494	0.036	0.470	0.383	0.087	0.499	0.408	0.090
				(7.63)	(3.84)	(0.25)	(6.02)	(2.67)	(0.56)	(6.28)	(2.70)	(0.58)
UMD							-0.160	-0.118	-0.042	-0.163	-0.120	-0.043
							(-2.55)	(-0.92)	(-0.40)	(-2.98)	(-0.97)	(-0.41)
STR							0.101	-0.033	0.134	0.089	-0.043	0.132
							(0.73)	(-0.17)	(1.06)	(0.63)	(-0.22)	(1.03)
LTR							-0.002	0.180	-0.182	-0.041	0.145	-0.186
							(-0.03)	(1.73)	(-1.51)	(-0.55)	(1.40)	(-1.54)
LIQ										-0.111	-0.094	-0.017
										(-2.43)	(-1.76)	(-0.45)
Adj. R^2	-0.005	-0.003	-0.006	0.314	0.243	-0.017	0.353	0.258	-0.002	0.371	0.268	-0.009
Quarters	168	168	168	168	168	168	168	168	168	168	168	168

Table IX Predictive Regression Estimates: Occupational Risk

This table reports coefficient estimates from regressions of the one-month-ahead HML return on occupation-specific IHD estimates and a set of factor predictors, macroeconomic controls, and value factor predictors described in Table III. In column (1), we compute IHD using the real annual income growth series of electrical engineers. In column (2), we compute IHD using the real annual income growth series of auto mechanics. Both annual income growth series are constructed using data from the Current Population Survey, following Davis and Willen (2000). The estimation period is from 1977 to 2011. *t*-statistics computed using Newey and West (1987) adjusted standard errors are reported in parentheses below coefficient estimates.

Predictor, Month t	Electrical Engineers (1)	Auto Mechanics (2)
IHD	-0.095	0.054
DIGH DEMUDI	(-1.92)	(0.78)
RISK-RETURN	-0.204	-0.142
	(-1.11)	(-0.76)
DIV	2.563	2.665
	(1.10)	(1.15)
YLD	-0.578	-0.049
	(-0.71)	(-0.07)
TERM	0.901	1.474
	(1.16)	(2.28)
DEF	0.019	1.004
	(0.02)	(0.61)
Ind Prod Growth	0.824	0.586
	(1.36)	(1.07)
Inflation Rate	2.334	2.257
	(4.52)	(3.00)
LEI Index	0.046	0.158
	(0.41)	(1.47)
Unexp GDP Growth	0.016	0.097
1	(0.07)	(0.33)
Value Spread	8.861	0.279
	(0.80)	(0.02)
Small Value Spread	-2.288	3.432
I I I I I I I I I I I I I I I I I I I	(-0.23)	(0.28)
Value DivYld – Growth DivYld	1.213	1.471
	(0.96)	(0.98)
1m HML Return	-0.021	0.034
	(-0.15)	(0.28)
Constant	-19 243	-31 517
	(-0.99)	(-1.76)
Adi B^2	0.303	0.997
Voors	25	0.407
	ออ	อย

onward at an annual frequency, we end up with annual IHD estimates from 1977 to 2011 for each occupation. We use these new hedging demand estimates to estimate our baseline 1-month-ahead regressions. We report the results in Table IX.

Given the findings in Davis and Willen (2000), we expect that electrical engineers, who have the most to gain from income hedging, should have an IHD that more closely replicates our main result than that of auto mechanics, who are unlikely to exhibit income hedging behavior. Consistent with this conjecture, we find that electrical engineers' IHD is a statistically significant predictor of HML returns, with a coefficient of -0.095 (*t*-statistic = -1.92). In contrast, auto mechanics' IHD is a statistically insignificant predictor of HML (coefficient = 0.054, *t*-statistic = 0.78).

B. Top Income Earners

Recent work by Guvenen et al. (2017) suggests that income hedging motives vary across households due to cross-sectional differences in exposure to aggregate risk. The authors show that those at the very bottom and very top of the income distribution are the ones who bear income risk, with respect to both aggregate GDP and their employer's earnings. In contrast, wage and salary earners with less income risk occupy the middle of the income distribution. Given that most market participants are at the top of the income distribution (Campbell et al. (2016)), the findings in Guvenen et al. (2017) suggest that the IHD of the top earners should have the most predictive power for returns.

We test this conjecture by examining how IHD measures stratified across income quantiles perform as HML predictors. Specifically, we use the Piketty and Saez (2003) top income growth data to calculate annual IHD measures for the bottom 90%, 91% to 95%, 96% to 99%, and top 1% of households in the income distribution. Since these data are available only at an annual frequency, we end up with 42 observations spanning the period 1970 to 2011. We substitute these group-specific IHD measures in our predictive HML regressions and report the results in Table X.

We find that the most important IHD predictors are those calculated using income growth for the 96% to 99% and top 1% of households. The respective coefficients for these groups are -0.208 (*t*-statistic = -2.25) and -0.187 (*t*-statistic = -1.93). In contrast, the IHD coefficients for the bottom 90% and 91% to 95% of households are economically smaller (-0.124 and -0.089, respectively) and insignificant (*t*-statistics of -1.48 and -1.08, respectively).

Given that the IHD estimates are negative on average (Table II), our results suggest that high income households should prefer growth over value stocks. In a recent study, Betermier, Calvet, and Sodini (2017) find that relatively young investors (i.e., between the ages of 35 and 55) favor growth stocks to value. We link the two sets of results by showing that relatively younger households are also the ones with the highest labor income. For this analysis, we extract the distribution of average labor income across age groups using age and income data from the CPS for the 1980 to 2017 period. We use the CPS because our IRS-based data set does not have demographic information. We find that the top earners are indeed relatively young (i.e., between the ages of 45 and 59), while older individuals (i.e., older than 70) belong mostly to the bottom of the income distribution.

Table X Predictive Regression Estimates: Top Income Earners

This table reports coefficient estimates from regressions of the one-month-ahead HML return on income quantile-specific IHD estimates and a set of factor predictors, macroeconomic controls, and value factor predictors described in Table III. In columns (1) through (4), we compute IHD using the real annual income growth series of the top 1%, 96% to 99%, 91% to 95%, and bottom 90% of earners, respectively. These annual income growth series are from Piketty and Saez (2003). The estimation period is from 1970 to 2011. *t*-statistics computed using Newey and West (1987) adjusted standard errors are reported in parentheses below coefficient estimates.

Predictor, Month t	Top 1% (1)	96-99% (2)	91-95% (3)	Bottom 90% (4)
IHD	-0.187	-0.208	-0.089	-0 124
	(-1.93)	(-2.25)	(-1.08)	(-1.48)
RISK-RETURN	-0.133	-0.098	-0.102	-0.123
	(-0.92)	(-0.67)	(-0.64)	(-0.83)
DIV	-0.457	-0.370	0.343	0.037
	(-0.32)	(-0.25)	(0.21)	(0.02)
YLD	-0.092	-0.054	-0.235	-0.203
	(-0.22)	(-0.13)	(-0.50)	(-0.47)
TERM	1.214	1.287	1.009	1.072
	(2.55)	(2.61)	(1.90)	(2.18)
DEF	0.693	0.885	0.831	0.809
	(0.59)	(0.76)	(0.74)	(0.73)
Ind Prod Growth	-0.011	0.037	0.029	0.022
	(-0.03)	(0.11)	(0.08)	(0.06)
Inflation Rate	2.161	2.193	2.339	2.280
	(3.83)	(4.17)	(5.14)	(4.44)
LEI Index	-0.002	0.006	0.010	0.008
	(-0.02)	(0.09)	(0.13)	(0.11)
Unexp GDP Growth	0.302	0.293	0.305	0.301
	(1.65)	(1.58)	(1.79)	(1.71)
Value Spread	-2.590	-3.568	1.614	-0.605
	(-0.43)	(-0.57)	(0.26)	(-0.10)
Small Value Spread	0.386	1.718	-1.726	-0.582
	(0.07)	(0.31)	(-0.28)	(-0.10)
Value DivYld – Growth DivYld	1.443	1.422	1.263	1.339
	(1.47)	(1.49)	(1.23)	(1.33)
1m HML Return	0.067	0.066	0.054	0.065
	(0.42)	(0.42)	(0.35)	(0.41)
Constant	0.387	-1.938	-4.481	-2.298
	(0.04)	(-0.18)	(-0.36)	(-0.20)
Adj. R^2	0.390	0.405	0.345	0.361
Years	42	42	42	42

C. Labor Rigidities and Income Hedging

Recent literature suggests that income hedging motives might be less important because many firms insure their workers through labor contracts. Because such labor contracts shift some risk to stockholders, they end up affecting asset prices (Berk and Walden (2013), Marfe (2017)). We complement this recent work and argue that both hedging via labor contracts and financial markets can coexist.

To begin, it is important to consider the asset pricing implications of income hedging, since there is evidence that households consider income risk when making asset allocation decisions (e.g., Bonaparte, Korniotis, and Kumar (2014) and Betermier, Calvet, and Sodini (2017)). Guvenen et al. (2017) show that those at the very bottom and very top of the income distribution are the ones who bear income risk. In contrast, workers in the middle of the income distribution are more likely to be insured by their companies via labor contracts and do not bear much income risk. Consistent with this view, we show above that our results are driven mostly by the income risk of those at the top, and not the middle, of the income distribution.

Next, we present two additional tests that suggest hedging via labor contracts and financial markets can coexist. First, we consider the cross-sectional implications of firms insuring workers through labor contracts using the degree of unionization across states. Since unions typically advocate for rigid wage contracts (e.g., Shister (1943)), workers in areas with higher unionization rates should have a weaker motivation to self-insure in financial markets. To test this conjecture, we collect data on state-level unionization rates between 2000 and 2011. For each state, we calculate the average proportion of workers in the state who are union members. We then split our state-level panel into states with above- and below-median union representation. We estimate our baseline state-level regression with these two samples and report the results in Table XI.

We find that our main results hold in both subsamples. Specifically, in the presence of all baseline control variables, we find that IHD has a loading of -0.087 (*t*-statistic = -2.21) in the high-union-representation states and -0.125 (*t*-statistic = -2.09) in the low-union-representation states. Also consistent with the labor rigidity story, the IHD estimate is larger in the low-unionization sample. However, the difference between the two coefficients is not significant (Wald test *p*-value = 0.593).

In our second test, we compute the labor share measure of Marfè (2017) and include it as an additional control variable in our national-level HML predictive regressions. We report these expanded regressions in column (7) of Table XI. We find that in this expanded regression, IHD remains an economically and statistically significant HML predictor, with a coefficient of -0.170 (*t*-statistic = -2.34).

Taken together, our findings suggest that our main results are not subsumed by potential wage rigidities from firms insuring workers through their labor contracts. Instead, the findings of Guvenen et al. (2017) suggest that the results of Berk and Walden (2013) and Marfè (2017) are likely driven by firms insuring the income streams of workers in the middle of the income distribution. In contrast, our results seem to be driven by the income hedging motives of those at the top of the income distribution, a group that is subject to significant income risk.

Predictive Regressions: State-Level Panel Union Representation and Aggregate Labor Share

This table reports estimates from within-state panel regressions of the one-month-ahead state-level HML return on state-level IHD measures. The state-level HML return is described in Table VI. The state-level IHD is based on conditional covariances between quarterly state income growth and national HML returns. In columns (1) to (3) ((4) to (6)), the sample consists of firms headquartered in states with above- (below-) median worker unionization rates. The estimation period is from Q1 1985 to Q4 2011. In column (7), the dependent variable is aggregate HML and we include the labor share of Marfe (2017) as an additional control. In column (7), the estimation period is from Q1 1970 to Q4 2011. t-statistics are reported below coefficient estimates. In columns (1) to (6), t-statistics are computed using standard errors adjusted for clustering at the state level. In column (7), *t*-statistics are computed using Newey and West (1987) adjusted standard errors.

	High	1 Union Represents	ation	Low	Union Representa	tion	Aggregate
${\rm Predictor,\ Month\ }t$	(1)	(2)	(3)	(4)	(5)	(9)	(2)
DHI	-0.085	-0.094	-0.087	-0.110	-0.128	-0.125	-0.170
	(-2.64)	(-2.54)	(-2.21)	(-2.13)	(-2.31)	(-2.09)	(-2.34)
RISK-RETURN	-0.082	-0.082	-0.085	-0.047	-0.050	-0.060	0.056
	(-2.95)	(-2.53)	(-2.63)	(-1.13)	(-0.99)	(-1.14)	(0.88)
DIV		0.178	1.326		0.736	0.970	-0.055
		(0.34)	(2.53)		(0.88)	(1.01)	(-0.06)
YLD		-0.518	-0.531		-0.357	-0.362	-0.015
		(-4.11)	(-3.96)		(-2.61)	(-2.24)	(-0.09)
TERM		-0.320	-0.002		-0.063	0.065	0.321
		(-1.77)	(-0.01)		(-0.21)	(0.18)	(1.02)
DEF		1.557	1.655		0.738	0.917	(-0.220)
		(2.32)	(2.23)		(1.13)	(1.30)	(-0.28)
Ind Prod Growth		-0.295	0.096		-0.270	-0.039	0.186
		(-1.74)	(0.63)		(-1.75)	(-0.23)	(0.62)
Inflation Rate		0.398	0.391		0.451	0.507	0.904
		(1.89)	(1.65)		(1.05)	(1.17)	(1.99)
							(Continued)

The Journal of Finance®

		Tabl	le XI-Continuea				
	High	ı Union Represer	itation	Lov	v Union Represen	tation	Aggregate
Predictor, Month t	(1)	(2)	(3)	(4)	(2)	(9)	(2)
LEI Index		0.011	0.073		0.047	0.071	-0.008
		(0.64)	(3.35)		(1.89)	(1.78)	(-0.17)
Unexp GDP Growth		0.418	0.293		0.347	0.280	0.169
		(6.75)	(4.96)		(5.95)	(4.27)	(1.37)
Value Spread		7.948	6.802		8.949	8.362	14.332
		(2.35)	(1.95)		(2.81)	(2.66)	(2.10)
Small Value Spread		-2.012	0.757		1.935	3.023	-8.844
		(-0.90)	(0.28)		(1.02)	(1.21)	(-1.71)
Value DivYld – Growth DivYld		1.538	1.589		1.499	1.524	1.039
		(8.30)	(7.78)		(5.00)	(4.56)	(2.16)
1m HML Return		0.016	0.016		-0.021	-0.015	-0.028
		(0.42)	(0.45)		(-0.47)	(-0.31)	(-0.22)
State Average Firm Size			-0.714			-0.368	
			(-2.15)			(-0.74)	
State Max Firm Size			0.015			0.004	
			(3.73)			(1.10)	
State Housing Growth			0.074			0.044	
			(1.21)			(0.83)	
Labor Share							-0.341
							(-1.09)
Constant							10.020
							(0.48)
State Fixed Effects	Yes	Yes	Yes	Yes	Yes	Yes	
Adj. R^2	0.005	0.055	0.052	0.003	0.050	0.047	0.100
U.S. States	17	17	17	18	18	18	
State-Quarter Observations	2,523	2,523	2,301	1,962	1,962	1,829	168

Income Hedging, Dynamic Style Preferences, and Return Predictability 2099

D. Additional Robustness Checks

Finally, in the Internet Appendix, we conduct additional tests to verify the robustness of our results with respect to the estimation of our IHD measure and the inclusion of additional control variables. Specifically, in Table IA.II, we consider alternative estimation windows for the IHD measure (8 to 15 years). In Table IA.III, we construct the IHD measure using different assumptions regarding the set of financial assets that are available to investors. In Table IA.V, we exclude some periods in which HML returns were extremely high or low. In Table IA.VI, we change the methodology of estimating hedging demand and use innovations from a VAR system to construct the IHD variable. In all four cases, IHD is a significant predictor of HML returns.

Furthermore, in Table IA.VII, we augment our baseline specification to control for income skewness, market effects captured by the market betas, and income growth volatility. We find that in the presence of these additional control variables, IHD remains a strong predictor of HML returns. In Table IA.VIII, we decompose the HML IHD into its value and growth components. The value (growth) IHD component is negatively (positively) related to future HML returns. Moreover, the value (growth) component of the HML IHD is negatively related to the value (growth) component of HML returns. Last, in Table IA.IX, we consider the IHD measures for the market, SMB, and UMD factors. The market IHD predicts market returns in our full sample. However, the IHDs for the SMB and HML factors predict the respective factor returns only in the later part of our sample. Overall, the evidence consistently supports our main finding that hedging demand-based measures are significant predictors of asset returns.

VII. Summary and Conclusion

This study examines the relation between income hedging and asset prices. Specifically, we investigate whether shifts in asset demand that are generated by income hedging motives can influence the prices of financial assets that facilitate income hedging. We focus on the income hedging potential of value and growth stocks and the pricing of the value-minus-growth portfolio, that is, the HML factor.

Since hedging demands are not directly observable, we propose a new method for estimating such demands for financial assets. Our proposed measure is based on the equilibrium of an economy with workers and sophisticated investors. Specifically, our IHD is the negative value of the covariance between asset returns and income scaled by the variance-covariance matrix of asset returns. Using our model-implied measure of income hedging, we show that systematic shifts in asset demands induced by income hedging motives affect the pricing of the HML portfolio in the short run. In economic terms, a onestandard-deviation decrease in IHD is associated with a 0.59% higher HML return next month.

Income Hedging, Dynamic Style Preferences, and Return Predictability 2101

We next exploit the geographic variation in income risk across the United States and show that state-level HML returns can be predicted using statelevel estimates of hedging-induced demand shifts. A long-short portfolio that exploits this hedging-induced predictability earns an annualized risk-adjusted return of over 6% during the 1970 to 2011 sample period.

These findings can be extended along several dimensions. For tractability, our current study focuses on a set of standard financial portfolios, but our method can be generalized. Our hedging demand estimation framework can also be adapted to consider other types of risk such as real estate risk and entrepreneurial risk. In addition, it may be interesting to further exploit heterogeneity in income risk across different occupations and various demographic groups to obtain more accurate aggregate hedging demand estimates. We leave these interesting topics for future research.

> Initial submission: December 9, 2015; Accepted: September 19, 2017 Editors: Bruno Biais, Michael R. Roberts, and Kenneth J. Singleton

Appendix A: Proof of Proposition 1

Let W_t^k be initial wealth, $\lambda_{r,t}^k$ be an $n \times 1$ vector of holdings in the risky assets, which is expressed in number of shares, and $\lambda_{r_f,t}^k$ be holdings in the risk-free asset for $k \in \{wo(\text{rkers}), so(\text{phisticated investors})\}$. To prove Proposition 1, we first derive optimal asset demands for the two sets of investors. We then aggregate these demands imposing market clearing.

A.1. Sophisticated Investors

During the first period, sophisticated investors receive an endowment W_t^{so} that can be invested in the risky and risk-free assets. The first-period budget constraint for the sophisticated investors is

$$W_t^{so} = \lambda_{r_f,t}^{so} + \boldsymbol{\lambda}_t^{so'} \boldsymbol{p}_t + \frac{q}{2} \boldsymbol{\lambda}_t^{so'} \boldsymbol{\lambda}_t^{so},$$

where p_t denotes the vector of prices for the risky assets, and the last term captures the quadratic costs of investing in risky assets. By assumption, the price of the risk-free asset, whose supply is perfectly elastic, is one. The second-period budget constraint is

$$W_{t+1}^{so} = \lambda_{r_f,t}^{so}(1+r_f) + \boldsymbol{\lambda}_t^{so'}(\boldsymbol{p}_{t+1} + \boldsymbol{d}_{t+1}),$$

where r_f is the interest on the risk-free asset and d_{t+1} is the vector of dividends for the risky assets. Solving the first-period budget constraint for $\lambda_{r_f,t}^{so}$, we can rewrite the second-period budget constraint as

$$W_{t+1}^{so} = W_t^{so}(1+r_f) + \lambda_t^{so'} (\boldsymbol{p}_{t+1} + \boldsymbol{d}_{t+1} - (1+r_f)\boldsymbol{p}_t) - \frac{q}{2}(1+r_f)\lambda_t^{so'}\lambda_t^{so}.$$

Since there is no intermediate consumption, the risk-neutral sophisticated investors maximize second-period expected wealth $\mathbb{E}_t^{so}[W_{t+1}^{so}]$ w.r.t. λ_t^{so} . Following Gârleanu and Pedersen (2013), we define returns in excess of the risk-free rate as $\mathbf{r}_{t+1}^x = \mathbf{p}_{t+1} + \mathbf{d}_{t+1} - (1+r_f)\mathbf{p}_t$. Thus, optimal portfolios for sophisticated investors are $\lambda_t^{so} = \mathbb{E}_t^{so}[\mathbf{r}_{t+1}^x]/(q(1+r_f))$. The expectation $\mathbb{E}_t^{so}[]$ is sophisticated investors' rational expectations operator, which reflects complete information.

A.2. Workers

Workers do not pay transaction costs when investing in risky assets. In addition to their first period endowment $(W_t^{wo} = y_t)$, workers receive a second-period endowment y_{t+1} . Hence, the second-period budget constraint for workers is

$$W_{t+1}^{wo} = W_t^{wo}(1+r_f) + \lambda_t^{wo'} r_{t+1}^x + y_{t+1}^{wo}.$$
(A1)

Let $\mathbf{z}_{t+1} = [\mathbf{r}_{t+1}^{x'} \Delta y_{t+1}]'$ be the $(n+1) \times 1$ vector of excess returns and income in first-differences, with $\mu_{z,t} = \mathbb{E}_t[\mathbf{z}_{t+1}]$ and $\mathbf{\Sigma}_{z,t} = Var_t(\mathbf{z}_{t+1})$ the conditional mean and variance of \mathbf{z}_{t+1} , respectively. We assume that workers form expectations using Bayes's rule. Also, to derive closed-form solutions for asset demands, we assume that workers believe that excess returns and income in first-differences are i.i.d. normal random variables with unknown parameters. Finally, workers have uninformative priors, which are given by $p(\boldsymbol{\mu}_{z,t}, \mathbf{\Sigma}_{z,t}) \propto |\mathbf{\Sigma}_{z,0}|^{-\frac{n+2}{2}}$ for some positive-definite and symmetric $(n+1) \times (n+1)$ matrix $\mathbf{\Sigma}_{z,0}$.

Under these assumptions, Zelner (1971) shows that the posterior distributions for $\mu_{z,t}$ and $\Sigma_{z,t}$ are: $\mu_{z,t}|\Sigma_{z,t}, \mathbf{z}_t \sim N(\widehat{\mu}_{z,t}, \Sigma_{z,t}/T)$, and $\Sigma_{z,t}|\mathbf{z}_t \sim IW_{n+1}(T, T\widehat{\Sigma}_{z,t})$, where N() is the normal distribution and $IW_{n+1}()$ is the inverse Wishart distribution. Moreover, the estimates for the mean and variance of \mathbf{z}_{t+1} are given by the following expressions:

$$\widehat{\boldsymbol{\mu}}_{z,t} = \widehat{\mathbb{E}}_t[\boldsymbol{z}_{t+1}] = \frac{1}{T} \sum_{\tau=t-T+1}^t \boldsymbol{z}_{\tau}, \quad \widehat{\boldsymbol{\Sigma}}_{z,t} = \widehat{Var}_t(\boldsymbol{z}_{t+1}) = \frac{1}{T} \sum_{\tau=t-T+1}^t (\boldsymbol{z}_{\tau} - \widehat{\boldsymbol{\mu}}_{z,t})(\boldsymbol{z}_{\tau} - \widehat{\boldsymbol{\mu}}_{z,t})'.$$

As shown above, the mean and variance estimates are based on the T most recent observations. In other words, workers in period t use data from (t - T + 1) to t. The fact that workers use past observations to estimate return and income moments is an implication of the assumption that they are Bayesians with uninformed priors. Finally, according to Zelner (1971) and Klein and Bawa (1976), the standardized predictive distribution for \mathbf{z}_{t+1} , $p(\mathbf{z}_{t+1}|\mathbf{z}_t)$, is a multivariate t-distribution with T - n - 1 degrees of freedom and parameters $\hat{\boldsymbol{\mu}}_{z,t}$ and $\frac{T+1}{T-n-1} \hat{\boldsymbol{\Sigma}}_{z,t}$.

Workers are risk-averse and have identical quadratic preferences. We use quadratic utility so we can derive closed-form solutions for their asset demands. Following Avramov and Zhou (2010), the optimal Bayesian portfolio is obtained

by maximizing expected utility under the predictive distribution

$$\max_{\lambda_t^{wo}} \widetilde{\mathbb{E}}_t \big[W_{t+1}^{wo} \big] - \frac{\gamma}{2} \widetilde{Var}_t \big[W_{t+1}^{wo} \big], \tag{A2}$$

where $\widetilde{\mathbb{E}}_t$ and \widetilde{Var}_t are the mean and variance under the predictive *t*-distribution, and W_{t+1}^{wo} is workers' wealth from equation (A1). In this case, the solution for λ_t^{wo} in equation (A2) is

$$oldsymbol{\lambda}_t^{wo} = rac{T-n-3}{\gamma(T+1)} \widehat{oldsymbol{\Sigma}}_{r,t}^{-1} \widehat{oldsymbol{\mu}}_{r,t} - \widehat{oldsymbol{\Sigma}}_{r,t}^{-1} \widehat{oldsymbol{\Sigma}}_{y,r,t},$$

where $\widehat{\Sigma}_{r,t} = \widehat{Var}_t(\mathbf{r}_{t+1}^x)$ is the estimated covariance matrix of excess returns, $\widehat{\mu}_{r,t} = \widehat{\mathbb{E}}_t[\mathbf{r}_{t+1}^x]$ is the vector of the estimated risk premia, and $\widehat{\Sigma}_{y,r,t} = \widehat{Cov}_t(\mathbf{r}_{t+1}^x, \Delta y_{t+1}^{w_0})$ is the vector of the estimated covariances between excess returns and income in first-differences.

A.3. Market Clearing

At any point of time, the new generation of sophisticated investors and workers buys the assets owned by the old generation. Therefore, when markets clear, the sum of the population-weighted asset holdings of the new generation is equal to the vector of the outstanding number of shares \bar{N} , which is constant because risky assets are assumed to be in limited fixed supply. Formally, the market-clearing condition is given by $\bar{N} = (1 - \nu)\lambda_t^{so} + \nu\lambda_t^{wo}$. Replacing the expressions for optimal asset holdings, we get

$$\bar{\boldsymbol{N}} = \frac{1-\nu}{q(1+r_f)} \mathbb{E}_t^{so} \left[\boldsymbol{r}_{t+1}^{x} \right] + \frac{\nu(T-n-3)}{\gamma(T+1)} \widehat{\boldsymbol{\Sigma}}_{r,t}^{-1} \widehat{\boldsymbol{\mu}}_{r,t} - \nu \widehat{\boldsymbol{\Sigma}}_{r,t}^{-1} \widehat{\boldsymbol{\Sigma}}_{y,r,t}.$$

REFERENCES

- Acemoglu, Daron, Ufuk Akcigit, and Murat Alp Celik, 2014, Young, restless and creative: Openness to disruption and creative innovations, NBER Working Paper No. 19894.
- Amihud, Yakov, Clifford M. Hurvich, and Yi Wang, 2009, Multiple-predictor regressions: Hypothesis testing, *Review of Financial Studies* 22, 413–434.
- Angerer, Xiaohong, and Pok-Sang Lam, 2009, Income risk and portfolio choice: An empirical study, *Journal of Finance* 64, 1037–1055.
- Avramov, Doron, and Guofu Zhou, 2010, Bayesian portfolio analysis, Annual Review of Financial Economics 2, 25–47.
- Barber, Brad M., and Terrance Odean, 2000, Trading is hazardous to your wealth: The common stock investment performance of individual investors, *Journal of Finance* 55, 773–806.
- Barber, Brad M., and Terrance Odean, 2001, Boys will be boys: Gender, overconfidence, and common stock investment, *Quarterly Journal of Economics* 116, 261–292.
- Barberis, Nicholas, 2000, Investing for the long run when returns are predictable, *Journal of Finance* 55, 225–264.
- Barberis, Nicholas, and Andrei Shleifer, 2003, Style investing, *Journal of Financial Economics* 68, 161–199.

- Barberis, Nicholas, Andrei Shleifer, and Jeffrey Wurgler, 2005, Comovement, Journal of Financial Economics 75, 283–317.
- Berk, Jonathan B., Richard C. Green, and Vasant Naiko, 1999, Optimal investment, growth options, and security returns, *Journal of Finance* 54, 1553–1607.
- Berk, Jonathan B., and Johan Walden, 2013, Limited capital market participation and human capital risk, *Review of Asset Pricing Studies* 3, 1–37.
- Betermier, Sebastien, Laurent E. Calvet, and Paolo Sodini, 2017, Who are the value and growth investors? *Journal of Finance* 72, 5–46.
- Betermier, Sebastien, Thomas Jansson, Christine Parlour, and Johan Walden, 2012, Hedging labor income risk, *Journal of Financial Economics* 105, 622–639.
- Bonaparte, Yosef, George M. Korniotis, and Alok Kumar, 2014, Income hedging and portfolio decisions, *Journal of Financial Economics* 113, 300–324.
- Campbell, John Y., 1996, Understanding risk and return, *Journal of Political Economy* 104, 298– 345.
- Campbell, John Y., Tarun Ramadorai, and Benjamin Ranish, 2014, Getting better or feeling better? How equity investors respond to investment experience, NBER Working Paper No. 20000.
- Campbell, John Y., and Robert J. Shiller, 1988, The dividend-price ratio and expectations of future dividends and discount factors, *Review of Financial Studies* 1, 195–228.
- Campbell, John Y., and Luis M. Viceira, 1999, Consumption and portfolio decisions when expected returns are time varying, *Quarterly Journal of Economics* 114, 433–495.
- Campbell, John Y., and Tuomo Vuolteenaho, 2004, Bad beta, good beta, *American Economic Review* 94, 1249–1275.
- Campbell, Sean D., Stefanos Delikouras, Danling Jiang, and George M. Korniotis, 2016, The human capital that matters: Expected returns and high-income households, *Review of Financial Studies* 29, 2523–2563.
- Chevalier, Judith, and Glenn Ellison, 1997, Risk taking by mutual funds as a response to incentives, *Journal of Political Economy* 105, 1167–1200.
- Cocco, Joao F., Francisco Gomes, and Pascal J. Maenhout, 2005, Consumption and portfolio choice over the life cycle, *Review of Financial Studies* 18, 491–533.
- Cohen, Randolph B., Christopher Polk, and Tuomo Vuolteenaho, 2003, The value spread, *Journal* of *Finance* 58, 609–642.
- Constantinides, George M., and Darrell Duffie, 1996, Asset pricing with heterogeneous consumers, Journal of Political Economy 104, 219–240.
- Cooper, Michael J., Huseyin Gulen, and P. Raghavendra Rau, 2005, Changing names with style: Mutual fund name changes and their effects on fund flows, *Journal of Finance* 60, 2825–2858.
- Coval, Joshua D., and Tobias J. Moskowitz, 2001, The geography of investment: Informed trading and asset prices, *Journal of Political Economy* 109, 811–841.
- Cronqvist, Henrik, Stephan Siegel, and Frank Yu, 2015, Value versus growth investing: Why do different investors have different styles? *Journal of Financial Economics* 117, 333–349.
- Daniel, Kent, Mark Grinblatt, Sheridan Titman, and Russ Wermers, 1997, Measuring mutual fund performance with characteristic-based benchmarks, *Journal of Finance* 52, 1035–1058.
- Davis, Steven, and Paul Willen, 2000, Occupation-level income shocks and asset returns: Their covariance and implications for portfolio choice, NBER Working Paper No. 7905.
- De Long, Bradford J., Andrei Shleifer, Lawrence H. Summers, and Robert J. Waldmann, 1990, Noise trader risk in financial markets, *Journal of Political Economy* 98, 703–738.
- Fama, Eugene F., 1981, Stock returns, real activity, inflation, and money, American Economic Review 71, 545–565.
- Fama, Eugene F., and Kenneth R. French, 1988, Dividend yields and expected stock returns, Journal of Financial Economics 22, 3–25.
- Fama, Eugene F., and James D. MacBeth, 1973, Risk, return, and equilibrium: Empirical tests, Journal of Political Economy 81, 607–636.
- Fama, Eugene F., and G. William Schwert, 1977a, Asset returns and inflation, *Journal of Financial Economics* 5, 115–146.
- Fama, Eugene F., and G. William Schwert, 1977b, Human capital and capital market equilibrium, Journal of Financial Economics 4, 95–125.

- Frazzini, Andrea, and Owen A. Lamont, 2008, Dumb money: Mutual fund flows and the crosssection of stock returns, *Journal of Financial Economics* 88, 299–322.
- Garleânu, Nicolae, Leonid Kogan, and Stavros Panageas, 2012, Displacement risk and asset returns, Journal of Financial Economics 105, 491–510.
- Gârleanu, Nicolae, and Lasse Heje Pedersen, 2013, Dynamic trading with predictable returns and transaction costs, *Journal of Finance* 68, 2309–2340.
- Graham, Benjamin, and David L. Dodd, 1934, Security Analysis: Principles and Technique (McGraw-Hill, New York, NY).
- Guvenen, Fatih, Sam Schulhofer-Wohl, Jae Song, and Motohiro Yogo, 2017, Worker betas: Five facts about systematic earnings risk, NBER Working Paper No. 23163.
- Heaton, John, and Deborah Lucas, 2000, Portfolio choice and asset prices: The importance of entrepreneurial risk, *Journal of Finance* 55, 1163–1198.
- Jagannathan, Ravi, Keiichi Kubota, and Hitoshi Takehara, 1998, Relationship between laborincome risk and average return: Empirical evidence from the Japanese stock market, *Journal* of Business 71, 319–347.
- Jagannathan, Ravi, and Zhenyu Wang, 1996, The conditional CAPM and the cross-section of expected returns, *Journal of Finance* 51, 3–53.
- Jurek, Jakub W., and Luis M. Viceira, 2011, Optimal value and growth tilts in long-horizon portfolios, *Review of Finance* 15, 29–74.
- Keim, Donald B., and Robert F. Stambaugh, 1986, Predicting returns in the stock and bond markets, *Journal of Financial Economics* 17, 357–390.
- Klein, Roger W., and Vijay S. Bawa, 1976, The effect of estimation risk on optimal portfolio choice, *Journal of Financial Economics* 3, 215–231.
- Kogan, Leonid, Dimitris Papanikolaou, Amit Seru, and Noah Stoffman, 2017, Technological innovation, resource allocation, and growth, *Quarterly Journal of Economics* 132, 665–712.
- Kogan, Leonid, Dimitris Papanikolaou, and Noah Stoffman, 2017, Winners and losers: Creative destruction and the stock market, NBER Working Paper No. 18671.
- Koijen, Ralph S. J., Hanno N. Lustig, and Stijn Van Nieuwerburgh, 2017, The cross-section and time-series of stock and bond returns, *Journal of Monetary Economics* 88, 50–69.
- Korniotis, George M., 2008, Habit formation, incomplete markets, and the significance of regional risk for expected returns, *Review of Financial Studies* 21, 2139–2172.
- Korniotis, George M., and Alok Kumar, 2011, Do behavioral biases adversely affect the macroeconomy? *Review of Financial Studies* 24, 1513–1559.
- Korniotis, George M., and Alok Kumar, 2013, State-level business cycles and local return predictability, *Journal of Finance* 68, 1037–1096.
- Kumar, Alok, 2009, Dynamic style preferences of individual investors and stock returns, *Journal* of *Financial and Quantitative Analysis* 44, 607–640.
- Lee, Charles M. C., Andrei Shleifer, and Richard H. Thaler, 1991, Investor sentiment and the closed-end fund puzzle, *Journal of Finance* 46, 75–109.
- Lettau, Martin, and Sydney Ludvigson, 2001a, Consumption, aggregate wealth, and expected stock returns, *Journal of Finance* 56, 815–849.
- Lettau, Martin, and Sydney Ludvigson, 2001b, Resurrecting the (C)CAPM: A cross-sectional test when risk premia are time-varying, *Journal of Political Economy* 109, 1238–1287.
- Lewellen, Jonathan, and Jay Shanken, 2002, Learning, asset-pricing tests, and market efficiency, Journal of Finance 57, 1113–1145.
- Lustig, Hanno N., and Stijn Van Nieuwerburgh, 2005, Housing collateral, consumption insurance, and risk premia: An empirical perspective, *Journal of Finance* 60, 1167–1219.
- Mankiw, Gregory N., and Matthew D. Shapiro, 1985, Trends, random walks, and tests of the permanent income hypothesis, *Journal of Monetary Economics* 16, 165–174.
- Marfè, Roberto, 2017, Income insurance and the equilibrium term structure of equity, Journal of Finance 72, 2073–2130.
- Merton, Robert C., 1969, Lifetime portfolio selection under uncertainty: The continuous-time case, *Review of Economics and Statistics* 51, 247–257.
- Merton, Robert C., 1971, Optimum consumption and portfolio rules in a continuous-time model, Journal of Economic Theory 3, 373–413.

Merton, Robert C., 1973, An intertemporal capital asset pricing model, *Econometrica* 41, 867–887.

- Michaelides, Alexander, and Francisco Gomes, 2005, Optimal life-cycle asset allocation: Understanding the empirical evidence, *Journal of Finance* 60, 869–904.
- Newey, Whitney K., and Kenneth D. West, 1987, A simple, positive semi-definite, heteroskedasticity and autocorrelation consistent covariance matrix, *Econometrica* 55, 703–708.
- Pastor, Lubos, and Pietro Veronesi, 2009, Learning in financial markets, Annual Review of Financial Economics 1, 361–381.
- Petkova, Ralitsa, and Lu Zhang, 2005, Is value riskier than growth? Journal of Financial Economics 78, 187–202.
- Piketty, Thomas, and Emmanuel Saez, 2003, Income inequality in the United States, 1913-1998, *Quarterly Journal of Economics* 118, 1–39.
- Polkovnichenko, Valery, 2005, Household portfolio diversification: A case for rank-dependent preferences, *Review of Financial Studies* 18, 1467–1502.
- Samuelson, Paul A., 1969, Lifetime portfolio selection by dynamic stochastic programming, *Review* of *Economics and Statistics* 51, 239–246.
- Santos, Tano, and Pietro Veronesi, 2006, Labor income and predictable stock returns, *Review of Financial Studies* 19, 1–44.
- Shister, Joseph, 1943, The theory of union wage rigidity, *Quarterly Journal of Economics* 57, 522–542.
- Stambaugh, Robert F., 1999, Predictive regressions, *Journal of Financial Economics* 54, 375–421.
- Teo, Melvyn, and Sung-Jun Woo, 2004, Style effects in the cross-section of stock returns, *Journal* of *Financial Economics* 74, 367–398.
- Timmermann, Allan G., 1993, How learning in financial markets generates excess volatility and predictability in stock prices, *Quarterly Journal of Economics* 108, 1135–1145.
- Timmermann, Allan G., 1996, Excess volatility and predictability of stock prices in autoregressive dividend models with learning, *Review of Economic Studies* 63, 523–557.
- Van Nieuwerburgh, Stijn, and Laura Veldkamp, 2009, Information immobility and the home bias puzzle, *Journal of Finance* 64, 1187–1215.
- Viceira, Luis M., 2001, Optimal portfolio choice for long-horizon investors with nontradable labor income, *Journal of Finance* 56, 433–470.
- Vissing-Jorgensen, Annette, 2002, Towards an explanation of household portfolio choice heterogeneity: Nonfinancial income and participation cost structures, NBER Working Paper No. 8884.
- Wahal, Sunil, and Deniz M. Yavuz, 2013, Style investing, comovement and return predictability, *Journal of Financial Economics* 107, 136–154.
- Yogo, Motohiro, 2006, A consumption-based explanation of expected stock returns, *Journal of Finance* 61, 539–580.
- Zelner, Arnold, 1971, An Introduction to Bayesian Inference in Econometrics (Wiley, New York, NY).
- Zhang, Lu, 2004, The value premium, *Journal of Finance* 60, 67–103.

Supporting Information

Additional Supporting Information may be found in the online version of this article at the publisher's website:

Appendix S1: Internet Appendix.